Contents lists available at ScienceDirect



Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser



Applications of earth-to-air heat exchangers: A holistic review

Giouli Mihalakakou^a, Manolis Souliotis^{b,*}, Maria Papadaki^a, George Halkos^c, John Paravantis^d, Sofoklis Makridis^a, Spiros Papaefthimiou^e

^a Department of Environmental Engineering, University of Patras, Greece

^b Department of Chemical Engineering, University of Western Macedonia, Greece

^c Department of Economics, University of Thessaly, Greece

^d Department of International and European Studies, University of Piraeus, Greece

^e School of Production Engineering and Management, Technical University of Crete, Greece

ARTICLE INFO

Keywords: Earth-to-air heat exchangers Studies of EAHE Systems Experimental studies Hybrid EAHE Systems Economic assessment, contents

ABSTRACT

The building sector is responsible for 40% of primary energy consumption, with heating/cooling covering the most significant portion. Thus, passive heating/cooling applications have gained significant ground during the last three decades, with many research activities on the subject. Among passive cooling/heating applications, ground cooling (especially earth-to-air heat exchangers) has been highlighted as a remarkably attractive technological research subjects because of its significant contribution to the reduction of heating/cooling energy loads: the improvement of indoor thermal comfort conditions: and the amelioration of the urban environment. This paper presents a holistic review of state-of-the-art research, methodologies, and technologies of earth-to-air heat exchangers that help achieve energy conservation and thermal comfort in the built environment. The review covers the critical subject of the thermal performance of earth-to-air heat exchanger systems; experimental studies and applications; parametric studies for investigating the impact of their main characteristics on thermal efficiency; and recent advances and trends including hybrid technologies and systems. The models describing the thermal performance of earth-to-air heat exchangers systems were classified in numerical, analytical, and datadriven; their main theoretical principles were presented; and experimental validation was mentioned when carried out. System parameters were grouped into three categories: system design, soil types, and soil surface coverage. System design parameters, especially length and burial depth, bore the most important influence on the thermal efficiency of the system. The paper was rounded up with an economic assessment of system application, and the conclusions highlighted the need for more experimental work including laboratory simulators.

1. Introduction

Buildings are responsible for approximately 40% of global energy consumption and for one-third of total energy greenhouse gas emissions [1]. The biggest part of the energy consumed in the building sector is used for space heating and cooling, which cover almost one third of total energy demand [2]. The use of conventional systems for building heating/cooling requirements present a significant energy and environmental impact, such as increase of carbon dioxide (CO₂) emissions; global warming; aggravation of greenhouse and urban heat island effects; increase of peak electric load demands; degradation of indoor air quality, etc. [3]. During the last decades, the scientific and technical community has made a great effort to explore and investigate the most

energy efficient solutions, based on renewable energy sources (RES), for space heating/cooling which could simultaneously contribute to energy conservation and environmental protection [4–6]. A remarkable place among these solutions and measures is held by passive heating and cooling techniques and algorithms, which could offer a significant reduction of energy consumption, resulting in a potentially remarkable mitigation of the heat island effect and improvement of the urban microclimate [7].

Passive cooling/heating presented an impressive evolution during the last three decades, enriched with new research, techniques, and materials, and it can provide minimization of heating/cooling loads; excellent thermal comfort conditions; mitigation of the urban heat island; and an amelioration of the urban environment [8–10]. Passive heating/cooling techniques are mainly based on the three natural heat

* Corresponding author. E-mail address: msouliotis@uowm.gr (M. Souliotis).

E mai add cos. moothous@downi.gr (in: boanou

https://doi.org/10.1016/j.rser.2021.111921

Received 11 July 2021; Received in revised form 28 October 2021; Accepted 16 November 2021 Available online 20 November 2021 1364-0321/© 2021 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

List of a	bbreviation
ANN	Artificial Neural Network
BIPVT	Building Integrated Photovoltaic/Thermal
CFD	Computational Fluid Dynamics
COP	Coefficient of Performance
EATHE	Earth Air Tunnel Heat Exchanger
EAHE	Earth to Air Heat Exchanger
NZEBs	Near Zero Energy Buildings
PV	PhotoVoltaic
PVC	Polyvinyl Chloride
PVT	PhotoVoltaic Thermal
RE	Renewable Energy
RES	Renewable Energy Systems
VEAHE	Vertical Earth to Air Heat Exchanger

transfer mechanisms: conduction, convection and radiation while passive cooling uses extensively the following three natural heat sinks for heat dissipation: ground, sky, and water. The ground can act as a heat sink for cooling a building or an agricultural greenhouse, but also as a heat source for providing heating during the cold periods of a year. Ground cooling/heating is one of the most documented and widely applied passive methods based mainly on thermal properties and temperature distribution at the surface of the ground as well as below it, as ground temperature remains constant at a depth of 2.5–3 m throughout the year [11,12]. Ground cooling/heating includes the following two main strategies: (a) direct earth-coupling techniques with earth-sheltered buildings [13,14]; and (b) indirect earth-coupling techniques with earth-to-air heat exchangers [15–17].

Earth-to-air heat exchangers (EAHE) use ground as a heat sink/ source and consist of a single or multiple pipes buried in the ground, through which ambient or indoor air is circulated and heat is transferred from the air to the soil during the summer (cooling mode) or from the soil to the air during the winter (heating mode). The air at the outlet of the pipes is mixed with the indoor air of a building or an agricultural greenhouse [11,18–20]. The thermal performance of EAHE is strongly influenced by many parameters, which could be classified in three main categories: (a) system design parameters, such as pipe material, pipe length, pipe radius, burial depth, and number of pipes; (b) different soil types described and expressed by the thermophysical properties of the soil profile, such as moisture content, thermal conductivity, specific heat capacity, and thermal diffusivity; (c) environmental parameters responsible for the temperature distribution of the ground surface, such as short and long-wave radiation, convective heat transfer, wind speed, vegetation and ground surface coverage, and evaporation/condensation speed [1,18]. A single EAHE includes important environmental and heat transfer processes influencing the thermal performance of the system; these may be summarized as follows: (a) environmental and thermophysical processes responsible for the temperature distribution at the surface and various depths of the ground, such as short and long wave radiation, convective heat flux, ground surface coverage, evaporation, and wind speed; and (b) heat transfer mechanisms in the soil and pipe, including conduction in the soil and convection inside the pipe.

This paper aims to offer a holistic and critical perspective on the subject of EAHE systems by presenting a complete and integrated review of the following main research activities: (a) modeling, (b) experimental investigation, (c) parametric studies, (d) hybrid systems, and (e) economic assessment. Modeling the thermal performance of EAHE systems is an attractive subject encompassing many mainly deterministic models and a few data-driven ones. Deterministic models, which could be numerical [16,21–48] or analytical [49–60], are mainly based on heat and mass transfer processes in the air and in the soil, while data-driven models use historical data to train and test artificial intelligence

networks [61–63]. Moreover, a significant number of research studies describe experimental investigations and case studies of different EAHE systems, including open and closed systems, various climatic conditions, tube shapes, soil types and coverage. In addition, experiments have been carried out and presented to validate mathematical models, present innovative experimental configurations, and (mainly) estimate and assess the heating or cooling potential of the system [16,17,64–69].

Apart from modeling and experimental studies, one of the most interesting topics of EAHE research has been the investigation of the impact of main system design parameters (such as pipe material, pipe length, pipe diameter, burial depth, air speed inside the tube, and number of pipes) on the heating/cooling efficiency of the system. Another interesting topic has been the extensive study and analysis of a significant number of environmental parameters (such as soil type, and soil coverage) on the capacity of the system [36,70–74].

This paper also presents a review of recent EAHE system advancements, including combined and hybrid technologies to improve system efficiency, to minimize energy consumption, and provide good thermal comfort conditions. These hybrid technologies could include photovoltaic systems, integrated solar chimneys, solar and wind systems, coupled ventilated roofs, hydrogen technologies, buried pipes filled with phase change materials, etc. [1,75–93]. The presentation is rounded up with an economic assessment included in order to complete a holistic approach to the subject.

The main innovative features of this research may be summarized as follows:

- 1. It presents a holistic approach of earth-to-air heat exchangers, covering almost all aspects of an important subject in energy conservation, and a key factor influencing the major problem of climatic crisis, leading to energy savings and environmentally benign microclimate improvement.
- 2. It offers a critical perspective as regards earth-to-air heat exchangers modeling by classifying numerous models into deterministic and data-driven, and then classifying the deterministic into numerical and analytical. Moreover, this research offers a further classification of numerical models, which is derived from the way researchers describe the thermal processes of a system.
- 3. It also includes and expounds on a critical perspective on experimental studies and hybrid technologies, leading to interesting conclusions regarding the necessity of expanding and enhancing experimental investigations.

The manuscript is structured as follows. Section 2 covers modeling, with Section 2.1 focusing on numerical, Section 2.2 on analytical, and Section 2.3 on data driven models. Section 3 covers experimental while Section 4 covers parametric studies. Section 5 covers hybrid systems and Section 6 reviews economic studies. Finally, Section 7 presents the conclusions.

2. Modeling the thermal performance of earth to air heat exchangers

The ground temperature distribution and the energy balance at the ground surface are significant parameters for passive cooling/heating and especially for direct (earth-sheltered buildings) and indirect (EAHE systems) ground cooling/heating. The ground temperature at the surface and various depths within the soil profile has been investigated by both experimental and theoretical methods [13,15,53,89,94–107]. Modeling and assessing the thermal performance of an earth-to-air heat exchangers system requires a deep knowledge of ground temperature profiles. That means knowledge of daily and annual fluctuations of soil temperature at the surface of the ground and various depths below it. Taking into account mainly the crucial parameter of ground temperature distribution at the ground surface and various depths below it, many models for simulating and predicting the EAHE systems have been

developed and presented in the scientific literature [21–32,42–48]. Santamouris and Kolokotsa [7] classified models in (a) numerical and analytical deterministic models, where the entire problem is described and solved through a mathematical formulation, and (b) data-driven models, where predictions were achieved through training the proposed model with the appropriate set of historical data. In this research, the EAHE models are classified into three categories: (a) numerical [16, 21–48,73,74,108], (b) analytical [49–52,54–59,109], and (c) data-driven [61–63].

2.1. Modeling earth to air heat exchangers using numerical methods

Many numerical models have been developed and proposed during the last few decades for simulating and predicting the thermal performance of an EAHE system for heating/cooling of a building or of an agricultural greenhouse. These models are based on differential equations describing mainly the conductional and convectional heat transfer processes in the soil and inside the tubes respectively during the operation of the EAHE. The large majority of that type of models have adopted several approximations (e.g. geometries, initial and boundary conditions, etc.) for discretizing and solving the differential equation sets. According to Ref. [21], numerical models describing the thermal behavior of EAHE systems could be classified into two categories: (a) models where only a part of the proposed geometry is influenced by the pipe presence; and (b) models where the entire proposed geometry is considered influenced.

Numerical models are mainly based on conductional processes in the soil, caused by: (a) the ground surface boundary conditions described by the energy balance equation at the surface; and (b) the EAHE system's presence, as well as convection phenomena inside the tube. Some of the models which focus on a full description of heat transfer processes in the soil, mainly take into consideration only conductional processes. These models consider specific boundary conditions including the ground surface and the undisturbed ground and the pipe, while they also describe the convectional processes necessary for calculating the air temperature distribution inside the tube. Moreover, there are some numerical models which are focused on heat convectional phenomena regarding the air flow inside the tube, and employ computational fluid dynamics theory and modeling.

The following numerical models may be classified in the first category, including models with a complete and detailed description of heat transfer processes in the soil. Puri [32] proposed a transient model of a set of two differential equations expressing the coupled and simultaneous heat and moisture transfer in the soil for a single EAHE. He selected a one-dimensional axial geometry considering that soil temperature and moisture around the tube were constant initially, while at the boundary of a large axial distance, ground temperature distribution and moisture content were not influenced by the tube's presence. Thus, the basic heat and mass transfer equations, (1) and (2), were expressed as follows:

$$\rho_d C_s \frac{\partial T}{\partial t} = \frac{1}{r} \frac{\partial}{\partial r} \left(r k_s \frac{\partial T}{\partial r} \right) \tag{1}$$

$$\frac{\partial W}{\partial t} = \frac{1}{r\rho_d} \frac{\partial}{\partial r} \left(\left(r\rho_a D_s \frac{\partial w}{\partial T} \right) \frac{\partial T}{\partial r} \right) + \frac{1}{r} \frac{\partial}{\partial r} \left(rK \frac{\partial W}{\partial r} \right)$$
(2)

where *T* and *W* are soil temperature and soil moisture respectively, ρ_d and ρ_a are soil and air densities, C_s is the specific heat of wet soil, D_s is the soil vapor diffusion coefficient, k_s is the overall soil thermal conductivity, *K* is the soil moisture diffusivity, *r* is the polar coordinate, and *t* is the time variable.

Soil and air parameters such as soil thermal conductivity and diffusivity were taken into account and the system of the two differential equations with two independent variables (axial distance and time) was discretized numerically using the finite elements method. Those authors found that soil temperature profiles evolve at a faster rate than the corresponding moisture content.

The energy and mass balance equations of [32] were the basis for the development of a transient numerical model developed by Ref. [42]. This model considers one single EAHE, takes into account the simultaneous heat and mass transfer into the soil and pipe, and it is expressed in polar co-ordinates with three independent variables (axial distance, y-co-ordinate, and time) and two dependent (heat and moisture).

Thus, the heat and mass transfer differential equations, (3) and (4), were expressed as follows:

$$\rho_{d}C_{p}\frac{\partial T}{\partial t} = \frac{1}{r}\frac{\partial}{\partial r}\left(kr\frac{\partial T}{\partial r}\right) + \frac{\partial}{\partial y}\left(k\frac{\partial T}{\partial y}\right) - l_{g}\rho_{w}\frac{1}{r}\frac{\partial}{\partial r}\left(D_{U,vap}r\frac{\partial h}{\partial r}\right) - l_{g}\rho_{w}\frac{1}{r}\frac{\partial}{\partial y}\left(D_{U,vap}r\frac{\partial h}{\partial r}\right)$$
(3)
$$\frac{\partial h}{\partial y} = \left[\frac{1}{r}\frac{\partial}{\partial r}\left(D_{T}r\frac{\partial T}{\partial r}\right) + \frac{\partial}{\partial y}\left(D_{T}\frac{\partial T}{\partial y}\right)\right] + \left[\frac{1}{r}\frac{\partial}{\partial r}\left(D_{U}r\frac{\partial h}{\partial r}\right) + \frac{\partial}{\partial y}\left(D_{U}\frac{\partial h}{\partial y}\right)\right]$$

$$\left[T^{T}\frac{\partial r}{\partial r}\right]^{+}\frac{\partial y}{\partial y}\left(D_{T}\frac{\partial y}{\partial y}\right)\right]^{+}\left[r\frac{\partial r}{\partial r}\left(D_{U}r\frac{\partial r}{\partial r}\right)^{+}\frac{\partial y}{\partial y}\left(D_{U}\frac{\partial y}{\partial y}\right)\right]$$
(4)

where *T* and *h* are soil temperature and moisture respectively, *k* is the soil thermal conductivity, D_T is the thermal moisture diffusivity, D_U is the isothermal moisture diffusivity, $D_{U,vap}$ is the moisture isothermal diffusivity in vapor form, ρ is the soil density, ρ_w is the moisture density, *r* is the polar coordinate – indicating the radial distance from the tube axis, *y* is the polar coordinate – indicating the axial distance from the tube inlet, and *t* is the time variable.

Many thermophysical soil and pipe parameters were taken into account, while boundary conditions were set axially at a large distance from the pipe, where it was assumed that temperature and moisture profiles are not influenced by the pipe's presence and at the pipe vicinity, where heat transfer depends on heat losses from the pipe to the ground and moisture migration due to the exchanger is zero with the pipe considered impervious. The differential equations were discretized and solved using the numerical method of control-volume and the results were validated and found sufficiently accurate against extensive sets of experimental data. The model was extended to calculate the thermal performance of a system of multiple EAHEs using the mathematical algorithm of superposition [43].

Bojic et al. [44] developed a set of linear equations describing the thermal performance of an EAHE system for heating/cooling a building. The mathematical model of the EAHE system consists of steady-state heat balance equations applied in each soil layer, assuming that soil is divided in layers with uniform temperature values. Pipes are considered all on the same layer. EAHE system was coupled with a building and authors calculated its energy needs and the contribution of the EAHE system to these needs. As shown, the system, with its main configuration parameter values, can effectively cover a significant part of building's energy needs especially during summer.

A transient numerical model simulating the thermal behavior of an EAHE system used for energy conservation in an agricultural greenhouse was developed in Ref. [45]. That model was based on heat transfer differential equation, (5), describing the energy movement into the soil in three dimensions (x, y, z) described as follows:

$$C\frac{\partial T}{\partial t} = \frac{\partial}{\partial x}\left(k\frac{\partial T}{\partial x}\right) + \frac{\partial}{\partial y}\left(k\frac{\partial T}{\partial y}\right) + \frac{\partial}{\partial z}\left(k\frac{\partial T}{\partial z}\right) + S$$
(5)

where.

C is volumetric heat capacity.

- T is temperature within the soil
- t is time variable
- *k* is heat conductivity.
- S is source term
- x, y, z are space coordinates.

A number of assumptions were considered mostly regarding soil thermal properties. The soil moisture gradient was considered not to affect the temperature gradient while the model took into account thermal processes and evaporation/condensation inside pipes. The numerical method of finite difference was selected for discretization and the model was successfully validated against experimental data.

A mathematical model combining numerical and analytical solutions was presented in Ref. [46]. The model calculated the soil and air temperature distributions using the form of Fourier integrals that correspond to an EAHE system being used for ventilation purposes. One-dimensional heat conduction and convection equations were used for calculating soil and air temperature inside the tube respectively. The ground temperature profile was estimated as a superposition of the undisturbed temperature profile and the temperature profile caused by the presence of the exchanger. The model was validated experimentally.

A two-dimensional set of heat and mass transfer differential equations describing thermal processes during the operation of a single earth tube for space heating, was solved numerically in Ref. [47]. Heat transfer was expressed mathematically by equation (6) as follows:

$$(\rho c)_{i} \frac{\partial T_{i}}{\partial t} = \frac{\partial}{\partial x_{i}} \left(\lambda_{i} \frac{\partial T_{i}}{\partial x_{i}} \right) + \frac{\partial}{\partial y_{i}} \left(\lambda_{i} \frac{\partial T_{i}}{\partial y_{i}} \right) + q_{i,z} + q_{i,p} \ (i = 1, ..., 22)$$
(6)

where ρ , c, and λ are the soil density, specific heat capacity, and thermal conductivity respectively, T_i is the ground temperature, x_i and y_i are space coordinates, $q_{i,z}$ and $q_{i,p}$ are heat quantities, and t is the time variable.

Boundary conditions were specified as the undisturbed ground temperature at a large distance from the pipe and as the energy balance equation at the ground surface, including an extensive number of parameters such as solar radiation, long wave radiation, convective and latent heat fluxes at ground surface, etc. All used parameters were defined while thermal balance equations and convection theory were used for estimating the air temperature distribution inside the pipe. The proposed model was developed and integrated inside an existing computing environment developed to simulate passive systems for heating in buildings. A parametric study was also presented.

Yoon et al. [48] presented a transient numerical method based on heat conduction equations in the soil, and taking into account the energy balance equation at the soil surface as well as the pipe presence for describing the thermal behavior of a system with multiple buried pipes used for space heating/cooling. The heat conduction equation in the soil was expressed by equation (7) as follows:

$$c_s \rho_s \frac{\partial \theta_s}{\partial t} = \lambda_s \left(\frac{\partial^2 \theta_s}{\partial x^2} + \frac{\partial^2 \theta_s}{\partial y^2} + \frac{\partial^2 \theta_s}{\partial z^2} \right)$$
(7)

where c_s is the soil specific heat capacity, ρ_s is the soil density, x, y, z are space coordinates, θ_s is the ground temperature, and t is the time variable.

A complete design description was offered for the multiple pipes and results were validated against measurements.

Tittelain et al. [21] proposed a transient numerical model for calculating the thermal performance of buried pipes. That model was based on discretizing the conductive flow using a response factor method. As regards geometry, a perpendicular to the pipes n-sections method was used to discretize the system. Validation was achieved by comparing theoretical results with those of two other experimentally validated models, one analytical and one numerical.

The computing environment of the EnergyPlus simulation program (https://energyplus.net/) was used to solve numerically the heat and mass transfer equations describing the thermal behavior of an EAHE system presented in Ref. [22]. The model took into account many parameters influencing the ground surface temperature profile, such as solar radiation, convection between ground and air, latent heat fluxes, etc. The results were successfully validated against measured values and the system proved efficient for energy conservation, able to provide 86% of the cooling load of July.

Trzaski and Zawada [23] proposed a quasi-three-dimensional, finite

element numerical model for heat transfer processes in soil and pipes, mainly based on the following conductive two dimensional heat transfer equation (8):

$$\frac{\partial}{\partial \tau} (c \cdot \rho \cdot T) = \frac{1}{\lambda} \left(\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} \right)$$
(8)

where, *c* is the soil heat specific heat capacity, ρ is the soil density, λ is the soil thermal conductivity, *T* is the temperature, *x* and *y* are space coordinates, and τ is the time variable.

The three dimensionality of the problem was achieved by dividing the soil area in separate cross-sections, thus producing a quasi-threedimensional model. The model includes an extensive number of parameters influencing the thermal behavior of the system, especially the air temperature at the pipe outlet. Those parameters were divided into three main categories: (a) system geometry (pipe length, burial depth, number of pipes, etc.); (b) condition of the ground surface (vegetation, ground shading, solar radiation, etc.); and (c) thermophysical parameters of the ground (density, moisture content, specific heat capacity, etc.). An extensive number of simulations were performed for different values of the aforementioned parameters, and their influence on the system efficiency was investigated. The results of the model were validated successfully against experimental data. A transient, threedimensional numerical model was also presented in Ref. [24], taking into account the coupled and simultaneous heat and mass transfer in the soil and pipe, during the performance of an EAHE system for cooling and pre-heating indoor spaces. A set of differential equations describing the mass and heat transfer in the soil was discretized and solved numerically using a control volume formulation. Boundary conditions were described analytically in all dimensions. The model results were compared successfully with the results of another model that was experimentally validated [110].

A two-dimensional, pseudo-transient numerical model describing the thermal performance of an EAHE system for space heating/cooling in three cities in Mexico, was developed and presented in Ref. [25]. The model was based on energy, mass, and momentum conservation equations, which were discretized and solved numerically with finite elements, using the restrictions of specific boundary conditions at the ground surface, at the pipe, and at a significant depth. Four Reynolds numbers were used for the simulations (100, 500, 1000, 1500) and the results showed remarkable potential for the heating/cooling system. Another two-dimensional, transient, numerical model was presented in Ref. [26], based on a combination of conductional and convectional heat transfer processes and using a control volume formulation for discretization. The profile estimation of the soil temperature was based on conduction theory, and the soil was divided in control volumes while boundary conditions were set at the soil surface, at the undisturbed ground, and at the tube. The convectional heat transfer process was taken into account for the air flow inside the tube, and Nusselt numbers were calculated for heating and cooling mode. Theoretical results were validated against existing experimental data and the model was found sufficiently accurate.

Fazlikhani et al. [27] and Belatrache et al. [28] offered methodologies for predicting the thermal performance and efficiency of EAHE systems in cold and hot arid climates in Iran and in arid climates in Algeria, respectively. Those methodologies were based on both conductive heat transfer in soil, formulating the soil temperature distribution at the surface and at various depths below it, and the convective heat transfer for the air flow inside the tube, thus calculating the air temperature inside the pipe. Both models were based on heat transfer theory, while differential equations were solved numerically and results were successfully validated against experimental data.

Rodrigues et al. [29] developed a transient, numerical model simulating the thermal performance of an EAHE system for different soil types, and for three different places in a coastal area of Brazil. Heat conduction equations were used for the soil temperature distribution, while the set of differential equations describing continuity, momentum, and energy was considered for the simulation of the air flow inside the tube, thus for the estimation of the air temperature inside the tube.

Liu et al. [30] developed a numerical model for predicting the thermal performance of a vertical EAHE system. Conductional and convectional heat transfer equations were used for estimating the distributions of the soil and air temperature. Three models were developed for air, tube, and soil, while a control volume formulation was used for the discretization process. Code was developed in MATLAB/Simulink environment, the results were compared with experimental data, and the model was found sufficiently accurate. The experimental system consisted of a U-shaped pipe (U-tube) made of stainless steel, 0.219 m in diameter, buried into a 16.5 m deep hole with a diameter of 1 m, filled with soil. Measurements were collected during both winter and summer. Specifically, Fig. 1 shows the comparison results between theoretical and experimental values for summer (1.a) and the winter (1.b) periods. As shown, there is good agreement between simulated and measured values for both summer and winter, where the relative deviation was 3.03% and 2.71% respectively and the average absolute error was less than 0.05 °C and 0.06 °C respectively.

A numerical, transient, two-dimensional model based on heat exchanged between the air flowing inside the tube and the soil was developed in Ref. [74]. That model took into account many soil thermal properties, and it was mainly used to investigate the impact of different soil coatings on the thermal performance of the system. Model results were validated experimentally (Figs. 2 and 3). The experimental system was an EAHE divided into three sections with different types of coating soils: sand, a mix of sand and bentonite, and initial earth. Each part is related to a vertical section in which temperature was measured. The length of the sand section was 10.40 m and the burial depth was 0.73 m. In addition, the length of the sand-bentonite was 10.40 m and the burial depth was 0.92 m while the length of the earth section was 8.2 m and the burial depth was 1.2 m. As shown, theoretical values were in good agreement with experimental ones for the months of January 2014 (Fig. 2) and July 2014 (Fig. 3). A maximum difference of 3.6 °C was observed between measured and simulated data (with a standard deviation of their differences equal 0.5 °C).

Attention now turns to research studies that included numerical models mainly based on convectional processes for different air flow patterns inside a tube. The most common characteristic of these models



Fig. 1. a. Comparison of measured and theoretical outlet air temperature values for the summer period [30]. Figure.b. Comparison of measured and theoretical outlet air temperature values for the winter period [30].



Fig. 2. Measured and calculated exit air temperature values for January 2014 (SD stands for standard deviation) [74].



Fig. 3. Measured and calculated exit air temperature values for July 2014 (SD stands for standard deviation [74].

is the extensive use of computational fluid dynamics (CFD) simulation software. Hollmuller and Lachal [31] developed a numerical model based mainly on heat and mass transfer processes between air and tube, which takes into account both sensible and latent heat fluxes and exchanges. For sensible fluxes, heat transfer was defined by the following equation (9), expressing the temperature difference between air and tube:

$$P_{sbl} = S_{tub}h(T_{air} - T_{tub}) \tag{9}$$

where.

 P_{sbl} is sensible air tube heat exchange.

 S_{tub} is total lateral heat exchange surface of tube

h is air/tube convective heat exchange.

 T_{air} is temperature of air.

 T_{tub} is temperature of tube.

The tubes were divided into nodes from inlet to outlet, and sensible heat was calculated using convective heat transfer equation, while latent heat was determined using water transfer equations for evaporation or condensation. Calculations of energy and water balance were performed for all tube nodes, and the heat transfer to soil nodes was calculated afterwards. Numerical results were extensively validated in four experimental sites, and the model was found sufficiently accurate.

The PHOENICS software [111] was used for the development of a transient, three-dimensional numerical model describing the thermal performance of an EAHE system in Ref. [33]. PHOENICS is an extensively used and documented CFD simulation program able to quantitatively predict fluid flows and mass transfer processes in many applications. Especially for buildings and outdoor spaces PHOENICS can estimate and predict the airflow in and around a building in order to improve energy efficiency, architectural design, and thermal comfort [111]. The numerical model used implicit methods based on simultaneous heat and mass transfer processes in the ground and pipe during the operation of the system, and it combined heat and mass transfer in the soil with convective fluid dynamic of turbulent flow inside the tube. Heat transfer in the soil was described by a three dimensional, transient, heat conduction equation while for modeling heat and mass transfer processes in the soil-air-tube system, two dimensional cylindrical co-ordinates were used. Thus, the heat transfer equation in the system soil-air-tube was expressed by equation (10) as follows:

$$\rho_a C_{p,a} \frac{\partial T_a}{\partial t} + div \left(\rho_a C_{p,a} u_a T_a \right) = div \left(\alpha_a \cdot grad T_a \right) + S_{\varphi}$$
⁽¹⁰⁾

where.

 ρ_a is density. $C_{p,a}$ is specific heat capacity. T_a is temperature of air in buried pipes u_a is velocity of air in buried pipes a_a is convective heat transfer capacity of air in buried pipe φ is general variable. S_{φ} is source item of the general variable t is time variable and div φ is defined by equation (11) as:

$$div\varphi = \frac{1}{r}\frac{\partial(r\cdot\varphi_r)}{\partial r} + \frac{\partial\varphi_x}{\partial x}$$
(11)

where.

r is coordinate.

The governing conservation of mass equation for the air in the buried pipes is given by equation (12):

$$\frac{\partial}{\partial t}(\rho_a u_a) + div(\rho_a u_a) = 0 \tag{12}$$

where.

 ρ_a is density

 u_a is velocity of air in buried pipes

t is time variable.

Model results were tested and found to be in close agreement with experimental data. CFD modeling, to predict and estimate the air flow and temperature variation inside the tubes using numerical methods was used in Ref. [34]. That research was focused on fluid dynamics inside the duct, based on convective heat transfer theory for turbulent flow. Mass conservation was expressed by the continuity equation (13):

$$\frac{\partial \rho}{\partial t} + \nabla(\rho \,\overrightarrow{\nu}) = S_m \tag{13}$$

where, ρ and ν are the fluid density and velocity while S_m is the fluid mass added or removed to the continuous or dispersed phase or any other source.

Furthermore, Navier-Stokes equations were used for describing fluid momentum conservation. Model results were validated against measured data while an artificial neural network (ANN) model was designed and trained for the estimation of convective heat transfer coefficients such as Nusselt numbers. Neural model results were compared and tested with those of the CFD model and found to be in close agreement.

Vaz et al. [16] offered a numerical solution of a set of differential equations consisting of (a) the Navier-Stokes equations describing the fluid mass flow conservation inside the tube; and (b) energy conservation equation as fluid also creates heat transfer inside the tube, thus calculating the temperature at each point of the tube. Model results were validated experimentally and the model was found to be accurate. Su et al. [35] developed two numerical sub-models for the description of a deeply buried EAHE system for space heating/cooling. The first one calculated air temperature values and moisture content inside the pipe, and was expressed by one-dimensional convective heat and mass transfer differential equations. The equations were discretized and solved numerically and implicitly, thus calculating the air temperature and moisture content at each of the defined node and finally at the pipe exit. The second sub-model described the ground temperature distribution by solving numerically the one-dimensional transient heat conduction equation in the soil, taking into account a large number of soil thermophysical parameters and using radial co-ordinate. Model results were tested with measurements and found to be in close agreement.

Sehli et al. [36] proposed a numerical, one-dimensional model to predict the thermal performance of an EAHE system for space heating/cooling. The model was mainly based on convective heat and momentum differential equation for turbulent flow, to estimate and predict the air temperature values at every point inside the tube. The convective parameters of Reynolds number and form factor were extensively used to simulate the system performance. The analytical solution of heat conduction equation at the ground surface and at various depths were used for the ground boundary conditions. Model results were successfully validated against experimental data.

A transient numerical model based on fluid mechanics with optimized geometry was developed by Ref. [37] to predict the thermal behavior of an EAHE system. Modeling used the convective flow differential equations for energy, mass, and momentum, while the one-dimensional conductive heat transfer equation in the soil was used for estimating the soil temperature profile. Model results were found to be in close agreement with literature data. Fluid dynamics modeling was also used to estimate the thermal performance of EAHE in the winter season for the hot and dry climates of India in Ref. [38]. The CFD model was based on the numerical solutions of three-dimensional Cartesian-coordinate differential equations expressing energy and mass transfer during the performance of an EAHE system. The set of continuity, momentum, and energy equations was solved numerically for specific boundary conditions especially at the tube surface and at the soil-pipe interface, while air temperature values were calculated at any point of the tube. Model results were successfully validated against experimental data.

A transient, numerical, three-dimensional simulation program based on CFD modeling was developed and presented in Ref. [73] for estimating the EAHE thermal performance for space cooling. The model was based on fluid mechanics theory and analysis, considering the fluid flow governing the differential equations of continuity, heat, and mass transfer. The model was solved numerically in the FLUENT environment, and its results were successfully validated against experimental data.

Ramirez-Davila et al. [39] developed a numerical model based on fluid mechanics to calculate the thermal performance of EAHE for three cities in Mexico. A finite volume formulation was used to discretize the three-dimensional differential equations describing energy, continuity, and momentum inside the pipe and the ground. The continuity, momentum, and energy conservation equations used in that model were expressed by equations (14)–(17) as follows:

$$\frac{\partial(\rho u)}{\partial x} + \frac{\partial(\rho v)}{\partial y} = 0$$
(14)

$$\frac{\partial(\rho uu)}{\partial x} + \frac{\partial(\rho vu)}{\partial y} = -\frac{\partial P}{\partial x} + \frac{\partial}{\partial x} \left[\mu \frac{\partial u}{\partial x} \right] + \frac{\partial}{\partial y} \left[\mu \frac{\partial u}{\partial y} \right]$$
(15)

$$\frac{\partial(\rho uv)}{\partial x} + \frac{\partial(\rho vv)}{\partial y} = -\frac{\partial P}{\partial y} + \frac{\partial}{\partial x} \left[\mu \frac{\partial v}{\partial x} \right] + \frac{\partial}{\partial y} \left[\mu \frac{\partial v}{\partial y} \right]$$
(16)

$$\frac{\partial(\rho uT)}{\partial x} + \frac{\partial(\rho vT)}{\partial y} = \frac{\partial}{\partial x} \left[\frac{\lambda}{C_P} \frac{\partial T}{\partial x} \right] + \frac{\partial}{\partial y} \left[\frac{\lambda}{C_P} \frac{\partial T}{\partial y} \right]$$
(17)

where.

P is pressure.

 ${\cal T}$ is temperature

 ρ is density

- λ is thermal conductivity
- *u*, *v* are horizontal and vertical velocities
- x, y are dimensional coordinates
- μ is viscosity.

Boundary conditions were set at the ground surface, where a great number of involved parameters were considered at the undisturbed ground temperature and at the pipe. For air temperature at the pipe inlet and outlet, convective heat transfer flow and computational fluid dynamics were used. Discretization was achieved using a CFD code and a finite elements methodology to calculate temperature values at any point inside the tube and at the exit, as a function of Reynolds number. The model was validated against measurements, and it was concluded that the system could be efficient for building heating and cooling.

Momentum, continuity, and energy differential equations were used in Ref. [26] as the governing equations for describing the thermal behavior of an EAHE system. The equations were solved numerically using a computational fluid dynamic piece of software and the finite volume method. That work also investigated the soil temperature recovery caused by the EAHE performance. The model results were successfully validated against experimental data.

Serageldin et al. [41] developed a transient mathematical formulation for predicting the thermal performance of an EAHE system for space heating/cooling and for the climate of Egypt. The model was based on a one-dimensional energy conservation equation, with a great number of assumptions concerning mainly the soil thermal condition and properties and the air flow inside the pipe. The energy conservation equation (including convection and diffusion processes) was expressed by equation (18) as follows:

$$\frac{\partial \rho T}{\partial t} + \frac{div(\rho Tu)}{Convection\ term} = \frac{div(k\ grad\ T)}{Diffusion\ term} + \frac{Q_{Air}^{Soil}}{Source\ term}$$
(18)

where.

 ρ is fluid density.

T is flowing air temperature

k is turbulence kinetic energy.

 Q_{Air}^{Soil} is heat flux from/to subsurface

u, *y*, *w* are velocity components in *x*, *y* and *z* direction

t is time variable.

A convectional three-dimensional model based on a CFD simulation software was also used to predict the air flow and temperature inside the pipe. The model results were extensively validated against experimental data.

Ahmed et al. [108] offered a CFD, two-dimensional, numerical model for describing the thermal performance of a horizontal EAHE system for space cooling. The model was based on the Navier-Stokes equations for momentum conservation, and on energy conservation equations, and it used the FLUENT simulation software [112] for solving the differential equations numerically. The model results were successfully validated against experimental data.

2.2. Modeling earth-to-air heat exchangers using analytical methods

Apart from numerical models, an extensive set of analytical models have been proposed in scientific literature. Most of them are based on heat and mass transfer processes during the EAHE operation and they take into account the analytical solution of heat conduction differential equation at the soil surface and various depths expressed by the following equation (19) [15]:

$$T_{z,t} = T_m - A_s \exp\left(-z\left(\frac{n}{365a}\right)^{0.5}\right) \cos\left[\frac{2\pi}{365}\left(t - t_0 - \frac{z}{2}\left(\frac{365}{\pi a}\right)^{0.5}\right)\right]$$
(19)

where.

 T_m is average annual earth temperature.

- A_s is temperature amplitude
- n is number of hours after sunset

a is lag coefficient

- z is depth
- t is time variable.

A simple, one-dimensional analytical model based on heat transferred inside a tube was developed and presented in Ref. [49] for predicting the thermal behavior of an earth-air tunnel for heating/cooling a hospital building in India. The model was based on the following energy balance equation (20) written for an infinitesimal element of the earth-air tunnel system in the direction of flow:

$$\dot{m}C_{a}\frac{\partial T_{a}}{\partial y}dy = \dot{Q}b \ dy \tag{20}$$

where.

- \dot{m} is mass flow rate of air.
- C_a is specific heat of air.
- T_a is tunnel air temperature.
- \dot{Q} is amount of heat transferred per unit area of tunnel surface
- b is perimeter of tunnel surface

y is position coordinate.

The model was validated against experimental data and found to be sufficiently accurate. Santamouris and Lefas [50] developed a one-dimensional analytical model for predicting the thermal behavior of an EAHE system for a hybrid agricultural greenhouse. The model was based on steady-state energy balance equations. The model was validated against existing measurements.

An analytical parametric model was implemented by Ref. [52] for estimating the thermal behavior of an EAHE system. The model was based on a systematic analytical parametric process developed and expressed using regression analysis. The following four parameters, influencing the thermal performance of the system, were used for the analysis: (a) pipe length; (b) pipe radius; (c) air velocity inside the tube; and (d) burial depth below the ground surface. The model results were successfully validated against accurate numerical data as well as experimental data, and the model was found accurate.

Krarti and Kreider [109] presented a simplified, transient analytical model for predicting EAHE thermal performance. The model, which was able to estimate the air temperature at the tube exit at any time, was based on heat transfer equations for conduction and convection, assuming that the system arrives at a quasi-steady state operation after a few days of performance. The model was successfully validated against measurements as shown in Fig. 4. A transient, analytical, three-dimensional model for predicting the thermal performance of an EAHE system was developed and presented in Ref. [54]. The model was based on the combination of heat flux and steady state equations. Authors did not offer experimental validation.

Deglin et al. [60] developed an analytical three-dimensional, transient model for predicting the thermal performance of a subsoil EAHE system for air-conditioning of a livestock buildings. The model was mainly based on convectional heat transfer processes between the air inside the tubes and the soil. The theoretical results were successfully validated against experimental data.

An analytical, steady-state, one-dimensional model was presented in Ref. [55] for predicting the air temperature distribution at the pipe exit of an EAHE system used for cooling a residential building in a desert



Fig. 4. Comparison of predicted with measured data [109].

climate. The model was based on an analytical solution of heat conduction equation in the soil, and calculating thermal resistances due to steady state thermal analysis and convective flow inside the tube. The convective heat transfer coefficient was calculated as a function of Reynolds and Nusselt numbers for laminar and turbulent flow. The model was successfully validated against experimental and theoretical parametric studies. Cucumo et al. [56] developed a transient, one-dimensional, analytical model for predicting the thermal behavior of an EAHE system. The model was based on heat conductional fluxes for a semi-infinite body as well as convective heat transfer, taking into account the latent heat flux inside the pipe. The model results were successfully validated against experimental data.

Lee et al. [57] offered an analytical transient module, developed inside the environment of the EnergyPlus simulation program, based on soil temperature distribution and thermal resistance equations due to convection phenomena caused by the tube presence. The model results were successfully validated against experimental and theoretical data. Do et al. [58] implemented an analytical, one-dimensional, steady state model to simulate a closed-loop EAHE for space cooling of residential buildings in hot and humid climates. Their model was simple and based on the analytical solution of conductive heat transfer equations in the soil as well as on thermal resistances of system components. The model was successfully validated against experimental data from published studies.

A transient, analytical approach describing the thermal behavior of an EAHE system was developed by Ref. [59]. The model was based on the harmonic temperature fluctuations of system components and of its thermal environments caused by conduction and convection processes. The model results were successfully validated against theoretical data received from validated numerical models. Rouag et al. [51] implemented a transient, one-dimensional, semi-analytical model for simulating the soil temperature profile around an EAHE system. That model was based on heat conduction processes in the soil in the pipe's vicinity. The model was successfully validated against literature results.

2.3. Modeling earth to air heat exchangers using data driven models

Data driven models are computing systems based on historical data

instead of the mathematical and physical formulations and equations of the deterministic models. Artificial Neural Networks (ANNs) models, which simulate biological neurons, and fuzzy logic models belong to this category of data driven models. The most important characteristic of the data driven models is their efficiency, in the sense that the analysis depends only on the available historical data, allowing the modeling of rather complicated non-linear systems and processes with sufficient accuracy [113]. Therefore, they often present significant advantages compared with deterministic models, including their simplicity as all they need is training and testing with historical data. Data driven models should be well trained to achieve an acceptable level of error between target and testing set of data, and then the system is ready for use.

Mihalakakou [61] presented an ANN approach for estimating the air temperature at the tube exit. In order to achieve the target three steps were considered: (a) developing a neural network architecture, (b) learning or training the network, and (c) testing the network. The network architecture consisted of a multiple layered feedforward network based on a backpropagation algorithm, which is a gradient descent of the mean square error as a function of weights [114]. If the mean square error exceeded some small predetermined value, a new cycle of training inputs presentation was started after the termination of the current one (epoch). One of the main parameters of the backpropagation algorithm is the learning rate, which specifies the size of changes made in the weights and biases at each epoch. The present ANN architecture consisted of one hidden layer of 16-21 log-sigmoid neurons followed by an output layer of one linear neuron with a learning rate of 0.5, while the number of epochs fluctuated between 3000 and 5000. The input set of data included the following parameters: (a) air temperature; (b) relative humidity; (c) ground surface temperature; and (d) ground temperature at burial depth, while the output parameter was the air temperature at the pipe exit. The model was extensively trained and tested and its results were successfully validated against the results of an accurate deterministic model, proving its efficiency in predicting the thermal behavior of an EAHE system.

A neural network model designed, trained, and tested to estimate air temperature distribution at the EAHE exit was presented in Ref. [62]. The neural network model had a typical multi-layered feed-forward architecture based on the back-propagation method, while a significant number of input parameters were selected. Thus, the hidden layer of ANN consisted of 15 tan-sigmoid neurons followed of one output layer of one linear neuron. A learning rate of 0.5 was selected while the number of epochs fluctuated between 100 and 300. Six input parameters influencing the EAHE thermal performance were used for training and testing the network: (1) pipe length, (2) ambient air humidity, (3) ambient air temperature, (4) ground surface temperature, (5) ground temperature at burial depth, and (6) air mass flow rate while the output was air temperature at the pipe exit. The results were extensively tested, compared with those of an experimentally validated numerical model [42], and the neural model was found accurate and appropriate for use.

Diaz et al. [63] developed both a thermo-dynamic model simulating the thermal performance of an EAHE system, and a fuzzy logic instead of an on/off controller, which could be coupled with the system in order to improve its energy efficiency. Their results showed that the fuzzy logic controller could find a desirable temperature and air mass flow rate instead of the on-off one, which always injected to the system the maximum air mass flow rate, resulting in the use of the fuzzy logic controlled reducing energy consumption for heating/cooling of a building by almost 88%.

A summary of the characteristics of the aforementioned models simulating the thermal performance of an EAHE system, numerical, analytical, and data driven is displayed in Table 1. This table is divided into three parts, one including the main characteristics of numerical model, a second with analytical models, and a third with data-driven ones. Each part contains the reference paper, a short description, and information on the validation mode (if it was carried out) of each presented model. As shown, most of numerical models were experimentally validated and were based on fundamental principles of heat and mass transfer in soil and air, with some of them giving importance to conductional processes in the soil and others to convectional ones in soil and air. Analytical models shown in the table were steady state or transient (including a simplified and a parametric one), while data driven models were mainly based on artificial neural network architecture.

3. Experimental studies

EAHE systems are used extensively to provide space heating/cooling and air-conditioning in buildings and agricultural greenhouses [20]. Many EAHE experimental studies have been carried out throughout the world. Trombe et al. [115] offered one of the first experimental studies of EAHE for space cooling in the south area of Toulouse (France). Two similar residential buildings were selected for the experiment, one of which was equipped with EAHE for comparative reasons. The main tube characteristics were: (a) PVC material; (b) diameter and thickness of 0.2 m and 5 mm respectively; (c) burial depth of 2.5 m; (d) length of 42 m; and (e) air flowrate equals to $0-45 \text{ m}^3/\text{h}$. Air temperature and relative humidity at the inlet and outlet of the pipe were experimentally measured parameters, with measurements taken every hour. The main result was that the air temperature at the pipe inlet fluctuated between 18 and 36 °C for a period of 8 days, while the air temperature variation at the pipe's outlet ranged between 18 and 25 °C, underscoring the potential of the system.

In [17] an EAHE system was used to provide air-conditioning in eight rooms of a double-floor guest house in India. A buried pipes system with forced air circulation was used in closed loop mode. The cooling potential of the system was investigated. Multiple heat exchangers were buried parallel to each other, with a distance of 1 m between adjacent pipes, at a depth of 2.5 m below the ground surface. The exchangers were made of concrete, with a length of 85 m, and a diameter of 0.5 m. The mean value of air velocity inside the tubes was 6.3 m/s. The ambient air temperature, the spatial air-flow velocity at the duct-openings, the spatial indoor air temperature, and the relative humidity of air-conditioned and non-air-conditioned rooms were measured every 2 hfor a month. The results showed that the ambient air temperature varied between 22.5 and 44.2 °C with a relative humidity between 9.4

Table 1

Main	characteristics of considered	models	simulating	the	thermal	performance	e
of an	EAHE system.						

NUMERICAL	MODELS	
[32]	Transient, based on coupled and simultaneous heat and moisture transfer in the soil.	Author did not mention any validation process
[42]	Transient, for a single pipe with three independent variables, based on coupled and simultaneous heat and moisture transfer into the soil caused by	Theoretical results were compared with experimental dat and model was found accurate
[43]	the tube's presence Transient, for multiple pipes, based on coupled and simultaneous heat and moisture transfer into the soil, it uses the mathematical method of	Model was successfully validated against an extensive set of experimental data and found accurate
[44]	Steady state heat balance	Authors did not mention any
[45]	Transient, three dimensional, based on heat transfer differential equations describing the energy movement into the soil in three	Model was successfully validated against experimental data and analytical predictions
[46]	dimensions (x,y,z) Transient, based on representation of temperature distribution in the form of Fourier integrals	Model was experimentally validated
[47]	Transient, two-dimensional, focused on heat conductional processes into the soil	Author did not mention any validation process
[21]	Transient, based on discretizing the conductive flow using a response factor method	Validation was achieved by comparing results with those of two other experimentally validated models, one analytical and one numerical
[48]	Transient, based on heat conduction equation in soil and taking into account the energy balance equation at the soil	Theoretical results were compared with experimental dat and the model was found accurat
[22]	Transient, takes into account a great number of parameters influencing the ground surface temperature profile such as solar radiation, convection between ground and air, latent heat fluxes, etc.	Model results were successfully validated against measured values and the system was prove efficient for energy conservation able of providing the 86% of cooling load in July.
[23]	Three-dimensional, finite element numerical model describing heat and mass transfer in soil and pipes including an extensive number of involving parameters influencing the thermal behavior of the system	Theoretical results were compared with experimental dat and the model was found accurat
[24]	Transient, three-dimensional, based on the coupled and simultaneous heat and mass	Model results were validated against the results of an experimentally validated model
[25]	Pseudo transient, two- dimensional, based on energy, mass, and momentum	Authors did not mention any validation process
[26]	Transient, two-dimensional, based on a combination of conduction and convectional heat transfer	Theoretical results were validate against existing experimental data and the model was found accurate
[27]	Transient, one-dimensional, based on both conductive heat transfer in the soil and on convective heat transfer in the air	Model results were successfully validated against experimental data.
	flow inside the tube	

G. Mihalakakou et al.

Table 1 (continued)

Reference	Short Description	Validation Mode
[00]	transfer in the soil and on convective heat transfer in the air flow inside the tube	Model results were successfully validated against experimental data.
[29]	based on conductional processes in the soil and on the set of differential equations describing continuity, momentum and	validation process
[30]	inside the tube Transient, based on conductional and convectional heat transfer equations used for estimating	Model results were compared with experimental data and the model was found accurate
[31]	both soil and air temperature distribution Steady state, based on sensible and latent heat fluxes and	The model was successfully validated against an extensive set
[22]	exchanges	of experimental data and it was found accurate
[33]	based on simultaneous heat and mass transfer, focused on convective processes in the pipe	experimental data and were in close agreement
[34]	Focused on fluid dynamics inside the duct and based on the convective heat transfer theory	Model results were successfully validated against measured data. An artificial neural network
	for turbulent flow	model was also designed and trained for the estimation of convective heat transfer
		coefficients. Neural model results were compared and tested with those of the CFD model and they were in close agreement
[16]	Transient, based on the Navier- Stokes equations describing the fluid mass flow conservation and on the energy conservation	Model results were validated experimentally and the model was found accurate
[35]	equation as the fluid flows inside the tube Transient consists of two sub-	Model results were tested with
[00]	models describing heat conduction and convection processes into the soil	measurements and they were in close agreement.
[36]	One-dimensional, based on convective heat and momentum differential equation for turbulent flow	Model results were successfully validated against experimental data
[37]	Transient, based on the convective flow differential equations for energy, mass, and	Model results were compared with literature data and they were in close agreement.
[38]	Transient, three-dimensional based on computational fluid	Model results were validated against experimental data and the
[39]	Two-dimensional, based on fluid mechanics and on energy,	Model was found accurate. Model was validated against measurements and it was
	continuity, and momentum inside the pipe and the ground	concluded that the system can be efficient for building heating and cooling.
[26]	Transient, based on computational fluid dynamics modeling	Model results were successfully validated against experimental data
[41]	Transient, one-dimensional, based on computational fluid dynamics modeling	Model results were successfully validated against experimental data
[108]	Transient, two-dimensional, based on computational fluid	Model results were successfully validated against experimental
[73]	Transient, three-dimensional, based on computational fluid	Model results were successfully validated against experimental
[74]	aynamics modeling Transient, two-dimensional, based on heat exchanged between the air flowing inside the	data Model results were successfully validated against experimental data
	pipe and the soil	with

Renewable and Sustainable Energy Reviews 155 (2022) 111921

Table	1	(continued)
Tubic		(communut)

-					
Reference	Short Description	Validation Mode			
ANALYTICAL MODELS					
[49]	Steady state, one-dimensional	The model was validated against experimental data and found accurate			
[50]	Steady state, one-dimensional	The model was successfully validated against existing			
[52]	Parametric model, regression analysis	Model results were successfully validated against accurate numerical data as well as against experimental data and the model			
[109]	Simplified, transient	Model results were successfully			
[54]	Transient, three-dimensional	Authors did not mention any			
[55]	Steady state, one-dimensional	Model results were successfully validated against experimental and theoretical parametric studies			
[56]	Transient, one-dimensional	Model results were successfully validated against experimental data			
[57]	Transient, inside the EnergyPlus environment	Model results were validated against experimental and theoretical data and found accurate.			
[58]	Steady state, one-dimensional	The model was successfully validated against experimental data from published experimental studies			
[59]	Transient, based on harmonic temperature fluctuations of system components.	Model results were successfully validated against simulation data from validated numerical models			
[51]	Transient, one-dimensional, semi-analytical	The model was successfully validated against literature results			
[60]	Transient, three-dimensional	The model was successfully validated against experimental data			
DATA DRIVEN MODELS					
[61]	Artificial Neural Network model based on back-propagation algorithm	The model was extensively trained and tested and its results were successfully validated against the results of an accurate deterministic model			
[62]	Artificial Neural Network model based on back-propagation algorithm	Results were extensively tested and compared with those of an experimentally validated numerical model and the neural model was found accurate			
[63]	EAHE system with a fuzzy logic controller	Authors did not mention any validation process			

and 75.8%; the air-conditioned room air temperature fluctuated between 25.3 and 28.4 °C, while the non-air-conditioned room air temperature was 2-5 °C higher. The indoor relative humidity was 40.7-48%for the non-air-conditioned room and 40.8-70.3% for the air-conditioned room, providing an acceptable level of comfort.

An investigation of the energy efficiency of several EAHE systems used for space heating and cooling in office buildings in Germany was presented in Ref. [64]. Three experimental projects were described, all in Germany. EAHE systems consisted of multiple (from 2 to 26) pipes, with lengths ranging from 67 to 107 m, diameters between 0.2 and 0.35 m, burial depths between 2 and 4 m, air velocity between 1.6 and 5.6 m/s, and soil dry-rocky, dry-gravel, and moist-clay soil types. The measurements period included one year for the three experimental projects. The experimental results could be summarized as follows: (a) the annual specific heating energy gain fluctuated between 16.2 and 51.3 kWh/m²; and (b) the annual specific cooling energy gain varied between 12.1 and 23.8 kWh/m².

Two experimental studies of EAHE systems used for providing

heating and cooling in agricultural greenhouses in Delhi (India) were presented in Refs. [66,100]. EAHE was designed and coupled with the greenhouse for heating during the winter, and cooling during the summer. The climate of the area is representative of tropical climates, with absolute minimum and mean minimum ambient air temperature values during the winter being around 4–9 °C respectively, and maximum and mean maximum air temperature values for the summer equal to 45 and 39 °C, respectively. The used EAHE system consisted of multiple PVC pipes, with a length of 39 m and a diameter of 0.06 m. The burial depth was 1.2 m below the ground surface, covered by bare soil. Pipes were arranged in the ground in a serpentine manner. The ground and indoor air temperature values were recorded, and it was concluded that the indoor air temperature increased by an average of 6–7 °C during the winter for [100], and up to 4 °C for [66]; while in the summer they decreased by 3–4 °C on the average for [100], and up to 8 °C for [66].

A solar greenhouse installed at Izmir (Turkey) was employed for an experimental study of an EAHE underground air tunnel in Ref. [116]. The experimental setup consisted of a horizontal U-bend type EAHE system (closed loop), including a galvanized pipe 47 m in length and 0.56 m in diameter, buried horizontally at a depth of 3 m, and a galvanized tube of 15 m in length and 0.8 m in diameter, coupled with the greenhouse, as illustrated in Fig. 5. The choice of different pipe diameters in the system employed by the authors of that work probably reflected an attempt to improve the efficiency of the system given any construction constrains. The soil was a combination of clay, sand, and small rock.

Those authors measured the temperature and humidity at many locations, including the system inlet and outlet, ambient space, indoor greenhouse, etc. The measured data were used to investigate the exergetic efficiency of the system, based on an energy and exergy analysis presented in the study. The overall efficiency of the exergy system efficiency was found to equal 60.7%.

Vaz et al. [16,117] developed and validated experimentally a numerical model describing the thermal performance of an EAHE system for space heating/cooling. The experimental study was conducted in a residential building in Brazil; the EAHE system consisted of three buried tubes, A, B, and C. All tubes were made of PVC, with tubes A and B being 0.11 m in diameter and buried at a depth of 2 m below the ground surface, and tube C being 0.10 m in diameter and buried at a depth of 2 m the temperature of the air flowing inside the tube had the potential to rise more than 8 °C or drop by about 4 °C. An extension of the experimental investigation described in Ref. [16], intended to assess the

heating/cooling potential of an EAHE system in the same residential building in Brazil, was performed in Ref. [117]. As in Ref. [16], the system consisted of three ducts, A, B, and C, all buried at different depths (1.60, 0.60, and 0.50 m correspondingly). Those authors found that the heating/cooling potential of the system was significant and could remarkably improve the thermal comfort conditions inside the building. Moreover, the results showed that the best periods for using the system were the months of February for cooling and May for heating.

An experimental ventilation system for space heating/cooling was used to validate a relevant numerical model [22]. The system consisted of six tubes, buried horizontally at a depth of 1.5–3 m below the ground surface, 50 m in length and 0.4 m in diameter. Experimental results may be summarized as follows: (a) the system could provide 62% of the heating requirements for the month of March, with a coefficient of performance (COP) of 3.2; (b) the system could provide 86% of the cooling load for the month of July, with a COP of 3.53.

Mongkon et al. [118] carried out an experimental investigation of a horizontal EAHE system to provide cooling in an agricultural greenhouse in Thailand. The EAHE system was buried at a depth of 1 m below a ground surface covered with a short lawn. The system consisted of an iron tube, placed in six serpentine shape rows, as illustrated in Fig. 6. Inlet and outlet tubes were made of PVC, and had a diameter of 0.08 m. Experimental results showed that the system had a significant cooling efficiency for tropical climates.

Chiesa et al. [67] presented a case study of monitoring the performance of an extensive EAHE system located in a school building in Imola (Italy), providing space heating and cooling, as shown in Fig. 7. The EAHE system consisted of three sets of pipes buried in three fields. Two sets had 12 pipes each, while the third set had only eight pipes. The pipes were 70 m in length and 0.25 m in diameter. The average burial depth below the ground surface was 2.61 m while the distance between adjacent pipes in the same field was 1.10 m. The average air velocity inside the tubes was 2 m/s. Those authors presented results of 12-month monitoring for all three fields of the EAHE system, which was proven remarkably effective for cooling and pre-heating modes.

Serageldin et al. [41] carried out an experimental study to validate their numerical model based on computational fluid dynamics. A 5.5 m long horizontal EAHE PVC pipe with a diameter of 0.05 m was buried in a serpentine shape at a depth of 2 m. Air circulated inside the tube at a velocity of 1–3.9 m/s while a loamy sand soil was selected. Experiments were performed in winter and summer, and the results showed a significant heating/cooling potential, which depended strongly on the design parameters of the system.



Fig. 5. Schematic diagram of experimental set up [116].



Fig. 6. Schematic diagram of experimental set up [118].



Fig. 7. Plan of installed EAHE system [67].

An earth-to-air heat exchanger system was used experimentally for cooling a residential building in Marrakesh (Morocco) [68]. The system consisted of three parallel U-shaped tubes, 77.7 m in length and 0.15 m in diameter, buried at a depth of 2.2–3.5 m below the ground surface.

Two of the pipes were coupled with the first floor of the building, while the third was connected with the second floor. The results showed remarkable system efficiency, offering an almost constant pipe outlet air temperature of 25 $^{\circ}$ C, when the ambient air temperature was over 40 $^{\circ}$ C.

An experimental investigation carried out to assess the cooling potential of an EAHE system for hot and arid climates, was presented in Ref. [119]. The experimental setup, depicted in Fig. 8, consisted of a pipe of 1.5 m in length and 0.15 m in diameter buried in soil contained in a galvanized steel drum. The system was connected with a blower and an air heater to circulate warm air inside the tube. The results showed that the system could be remarkably efficient, offering an outlet air temperature reduction of up to 24 $^{\circ}$ C.

Yusof et al. [120] employed an innovative laboratory simulator to assess the energy efficiency and the potential of an EAHE. The simulator acted as real ground surrounding the pipes, and allowed operation in steady state, eliminating the accumulation term in the energy balance equation. The EAHE laboratory simulator included an 8.7 m long and 101.8 mm inside diameter PVC pipe. The results showed that efficiency ranged up to 88%, while the air temperature reduction reached around 10 °C, thus achieving a temperature reduction of 27.5% compared with the air temperature at the pipe inlet. Liu et al. [30] carried out an experimental study to validate their transient numerical model for predicting the thermal performance of a vertical EAHE system. Experiment was performed in Changsha (China) which has a subtropical climate (hot summer and cold winter). The system consisted of a U-shaped pipe (U-tube), made of stainless steel, which was 0.219 m in diameter, buried into a 16.5 m deep hole with a 1 m diameter, filled with soil. Measurements were collected during both winter and summer. The authors were in favor of the Vertical EAHE (VEAHE), and experimental results were used to validate their model, so as to further study the parametric sensitivity of the EAHE system. As such, they studied the effect of tube material, tube insulation, air-flow, and soil types. Their results indicated that optimum choices depend on specific objectives. In



Fig. 8. Schematic diagram of experimental set up [119].

Ref. [69], an extension of experimental and theoretical research on VEAHE was presented and discussed. Those authors argues that U-shaped tubes presented two main advantages: (a) a significant reduction in their requirement of land use; and (b) an improvement of system efficiency due to increased burial depth, as shown in Fig. 9. The location of the experiment was again Changsha (China), and the experimental set up consisted of the U-tube buried in the hole at a depth of 16.5 m. Experimental results showed that the system can be very efficient in both summer and winter. The air temperature at the pipe outlet fluctuated between 22.4 and 24.4 °C in the summer, and between 16 and 18 °C in the winter. Moreover, VEAHE could exploit the quality of deep ground as a natural heat sink for the cooling period, as well as a natural heat source for the heating period of the year.

In [121] a full-scale experimental study was carried out to assess the cooling potential of an EAHE system in a city in southwest China, characterized by hot and humid climate. The experimental setup consisted of four buried PVC pipes of different design parameters (diameter of 0.16, 0.11, 0.11 and 0.075 m correspondingly). All pipes had a vertical inlet, a horizontal section where the heat transfer process was conducted, and a vertical outlet. The burial depth for three pipes was 3 m below the ground surface, and 2 m for the fourth one. Adjacent pipes buried at 3 m were 0.7 m apart. Experimental results showed that the EAHE system presented a remarkable cooling capacity and could significantly cool and dehumidify the ambient air. The outlet air temperature and humidity fluctuated between 21.5 and 41.2 $^{\circ}$ C,

and from 11.2 to 20.5 g/kg for an EAHE of a substantial length. The achieved maximum air temperature reduction for the buried pipe of 0.075 m in diameter buried at a depth of 3 m, was 22.1 °C, while for moisture content the reduction was 7.41 g/kg. Sensible heat transfer capacity varied in the range of 60–83%, while the cooling capacity for latent heat transfer fluctuated between 17 and 40%.

EAHE systems are multiparametric. The efficiency of the studied systems directly depended upon the environmental conditions and the engineering design, and their conclusions were relevant only for the specific conditions and design. Nevertheless, the aforementioned studies, shown in Table 2, were conducted in different places around the globe with different thermophysical properties and engineering choices. In spite of the fact that conclusions from individual research works cannot be extrapolated, there is some common ground and their findings converge to the following three facts: (1) they demonstrated that buried pipes are mostly affecting summer indoor temperatures (tropical cooling); (2) all system efficiency values fluctuated around 60%; and (3) the outlet air temperature did not guarantee a rise of the indoor temperature in the winter, but it appeared to efficiently reduce summer indoor temperatures. Thus, the above studies seem to suggest that EAHE are more efficient as a means of cooling rather than a method of heating.

Irrespective of the efficiency of those systems, they were not optimized on the needs of indoor air quality. Most research was conducted on empirically designed systems. Although the reported research has provided encouraging results which could find direct application without any further refinement in order to reduce energy requirements



A-A 1:2

Fig. 9. Schematic diagram of experimental set up [69].

EXPERIMENTAL STUDIES

EAHE

configuration

Material: PVC;

length: 42.0 m;

diameter: 0.2

m; depth: 2.5

m. The system

was buried

under the

building

garden

Multiple

material:

concrete: length: 85 m;

parallel tubes;

diameter: 0.5

6.2 m/s: tube spacing: 1 m

m; air velocity:

Multiple pipes;

lengths:

67–107 m;

diameters:

2-4 m: air velocities:

0.2–0.35 m;

burial depths:

1.6-5.6 m/s;

rocky, drygravel, and

moist clay.

diameter:

fashion

0.06; depth:

1.2 m; pipes

were arranged in a serpentine

Multiple pipes;

length: 39 m;

m; depth: 1.2

m; pipes were arranged in a

Horizontal U-

EAHE system

(closed loop);

length: 47 m;

diameter: 0.56

bend type

serpentine

fashion

diameter: 0.06

soil types: dry-

Multiple pipes;

length: 39 m;

parameters

Table 2

Reference

[115]

[17]

[<mark>64</mark>]

[65]

[66]

[116]

Summary of main characteristics and results of experimental studies systems.

Operation

60.7%

mode

Space

Space

Space

and

heating

cooling

Heating

cooling

Heating

cooling

Heating

cooling

and

and

and

cooling

cooling

Experimental

location and

Building type coupled with EAHE system

Residential

building in the

south of France

Double floor

India

guest house in

Office building in

Germany

Agricultural

greenhouse in

Delhi (India)

Agricultural

greenhouse in

Delhi (India)

Experimental

solar greenhouse

at Izmir (Turkey)

Penewahle and Sustainable Energy Paviaus 155 (2022) 111021

dies of EAHE EXPERIMENTAL STUDIES					
n Results	Reference	EAHE configuration parameters	Experimental location and Building type coupled with EAHE system	Operation mode	Main Results
		m; depth: 3 m; soil was a			
ir perature ation at the 's inlet: 36 °C ir	[16,117]	combination of clay, sand and gravel Multiple pipes with different design	Residential building in Brazil	Heating and cooling	Heating/ cooling potential of the
perature ation at the 's outlet: 25 °C mbient perature ation: -44.2 °C; ient tive ation: 75.8% oom indoor emperature ation:		parameters; material: PVC			system was significant and could improve the thermal comfort conditions inside the building remarkably. The best perioo for using the system was the month of February for cooling, and the month of
-28.4 °C; ior relative idity ation: -70.3% he specific ing energy fluctuated veen 16.2 51.3 kWh/ ber annum he specific ing energy fluctuated veen 12.1 23.8 kWh/ ber annum	[22]	Six horizontal tubes; length: 50 m; diameter: 0.4 m; depth: 1.5–3 m	Research Laboratory, University of Nottingham, Ningbo, China	Heating and cooling	May for heating Heating mode: system would provide 62% o the heating load for the month of March with a COP of 3.2. Cooling mode: system could provide 86% o the cooling load for the month of July with a COP of 3.53
oer annum oor air perature es eased by an age of °C during winter and veased by an age 3–4 °C the summer oor air perature eased by up	[118]	Horizontal tube, made of iron, placed in 6 serpentine shaped rows. Inlet and outlet tubes were made of PVC, 0.08 m in diameter. The system was buried under short-lawn ground	Agricultural greenhouse in Thailand (tropical climate)	Cooling	System presented significant efficiency for cooling in tropical climates
°C during winter le in the mer the use uried pipes lted in a perature ease hing 8 °C rall exergy em iency	[67]	surface. Three fields of buried pipes with different number of pipes (12, 12 and 8), 70 m in length and 0.25 m in diameter; depth: 2.61 m; air velocity: 2 m (c distance	School building in Imola, Italy	Heating and cooling	System proved remarkably effective for cooling and pre-heating modes

(continued on next page)

between

Table 2 (continued)

EXPERIMENTAL STUDIES				
Reference	EAHE configuration parameters	Experimental location and Building type coupled with EAHE system	Operation mode	Main Results
[41]	adjacent pipes: 1.10 m Horizontal EAHE in serpentine shape; material: PVC; inner diameter: 0.05 m; length: 5.5 m; air velocity: 1–3.9 m/s; soil: loamy	Energy Resources Engineering Department Laboratory, University of Science and Technology, Egypt	Heating and cooling	Significant heating/ cooling potential strongly depended on system design parameters
[68]	sand Three parallel U-shaped tubes; length: 77.7 m; diameter: 0.15 m; depth: 2.2–3.5 m	Residential building in Marrakesh (Morocco)	Cooling	System presented remarkable efficiency offering almost constant pipe outlet air temperature values of 25 °C when the ambient air temperature was higher than 40 °C
[119]	One pipe; length: 1.5 m; diameter: 0.15 m; system was connected with a blower and an air heater to circulate warm air inside the	Application for hot, dry, and arid climates	Cooling	40 °C System could be remarkably efficient, offering an outlet temperature reduction up to 24 °C
[120]	EAHE system laboratory simulator; pipe made of PVC; length: 8.7 m; inside diameter: 101.8 mm	Implementation in tropical climates	Cooling	System efficiency ranged up to 88% and the air temperature reduction reached 9.62 °C, achieving a temperature reduction of 27.5% compared with the air temperature at the pie in act
[30]	U-shaped pipe (U-tube) made of stainless steel; diameter: 219 mm, buried into a deep hole of 16.5 m depth and 1 m diameter, filled with soil	Experiment carried out in Changsha (China), a city of sub-tropical climate (hot summers and cold winters)	Heating and cooling	the pipe inlet. Experimental validation of numerical model
[69]	U-shaped tubes (U- tubes), as in [30]	Experiment carried out in Changsha (China), a city of	Heating and cooling	Air temperature at the pipe outlet fluctuated

Renewable and Sustainable Energy Reviews 155 (2022) 111921

EXPERIMENTAL STUDIES				
Reference	EAHE configuration parameters	Experimental location and Building type coupled with EAHE system	Operation mode	Main Results
		sub-tropical climate (hot summers and cold winters)		between 22.4 °C and 24.4 °C in the summer and 16.0 °C–18 °C in the winter
[121]	Four pipes made of PVC with different design parameters; depth: 3 m (three pipes), and 2 m (one pipe); . distance between adjacent pipes: 0.7 m	City in south- west China characterized by hot and humid climate	Cooling	EAHE system presented remarkable cooling capacity and was able to coo and dehumidif the ambient ai significantly. System cooling capacity as regards sensible heat transfer varied between 60.5% and 82.82%, while the cooling capacity for latent heat transfer fluctuated between 17.18% and

in buildings, the authors of this work consider that the need for research which probes into the science behind such applications is imperative. As such, it is recommended that research be conducted in laboratory simulators, where the effect of each parameter is evaluated in isolation or in groups. At the same time, fundamental experimental research on temperature profiles as a function of the distance from the earth surface, local climate, weather conditions, soil type, earth vegetation, and annual rainfall can be useful in selecting the most appropriate engineering design. Finally, a life cycle assessment is also necessary, e.g. the energy employed for the fans, digging the ground, manufacturing the tubes, etc., should not exceed the heating/cooling energy savings.

Table 2 offers a summary of the main characteristics and results of representative implementations of EAHE systems in various types of buildings, greenhouses and research laboratories.

4. Parametric studies

The research literature on EAHE systems includes many parametric studies expressed in the form of sensitivity analysis and investigation of the impact of the main system design and environmental parameters on the space cooling/heating efficiency. The thermal behavior of an EAHE is strongly affected by many parameters of different type and quality, which were classified into three main categories: (a) buried pipes system design parameters; (b) soil types described by the thermophysical characteristics of the ground; and (c) environmental parameters, mostly influencing the ground surface temperature distribution. Parameters in the first category include: pipe length; pipe radius; air velocity inside the tube; burial depth; number of pipes; and pipe material. Those in the second category include: soil moisture content; soil density; soil specific heat capacity; and soil thermal diffusivity. Finally, those in the third category include: short and long wave radiations; convective heat flux between air and ground surface; latent heat flux due to evaporation/ condensation; wind speed; vegetation; and different ground covers. Most researchers have investigated the effect of parameters in the first category (system design) on the cooling/heating capacity of EAHE systems, although there are also studies including parameters of the other two categories. Table 3 lists studies addressing the effect of crucial parameters on the thermal efficiency of an EAHE system.

Sodha et al. [70] investigated the effect of pipe length, radius, and air mass flow rate inside the tube, (first category); as well as earth surface coverage (third category), on the EAHE system efficiency for space cooling for hot and dry climates in India. Those authors presented the most effective solution as regards length, radius, mass flow rate, and number of pipes for the cooling period of the year. As regards different ground surface covers, sunlit, shaded, wetted, and wet shaded were considered, with wet shaded found to constitute the most effective solution.

Four main system design parameters, pipe length, pipe radius, air velocity inside the pipe, and burial depth have been regarded as keyvariables for extensive sensitivity analyses for a single pipe and for space cooling/heating in Refs. [122,123] respectively, and in Ref. [43] for multiple pipes and space cooling. The system cooling/heating capacity was shown to be strongly related to the previous design parameters. For the single pipe, an increase of pipe length and burial depth resulted in an increase of system efficiency, while an increase of pipe radius and air velocity resulted in a reduction of system capacity. For multiple pipes, the distance between adjacent pipes was also considered,

Table 3

Parametric studies of the effect of crucial parameters on the thermal efficiency of an EAHE system.

PARAMETRIC STUDIES			
Reference	Category of Parameters	Investigated Parameters	
[70]	(a), (c)	Pipe length, radius, air mass flow rate, ground surface cover	
[122]	(a)	Pipe length, radius, air velocity, burial depth	
[123]	(a)	Pipe length, radius, air velocity, burial depth	
[43]	(a)	Pipe length, radius, air velocity, burial depth, space between pipes	
[71]	(c), (a)	Ground surface cover, pipe length, radius, burial depth, air velocity	
[124]	(c), (a)	Ground surface cover, pipe length, radius, burial depth, air velocity	
[125]	(a)	Pipes length, pipes radius, burial depth, and air velocity inside the pipes	
[126]	(a)	Pipes length, pipes radius, burial depth, and air velocity inside the pipes	
[109]	(a)	pipe diameter, air velocity inside the tube	
[60]	(a), (b)	Pipe length, diameter, burial depth, air velocity, and soil types	
[47]	(a), (c)	Pipe diameter, burial depth, pipe material, and ground surface cover	
[57]	(a)	Pipe radius, length, air flow rate, and burial depth	
[127]	(a)	Burial depth, pipe length, air velocity inside the tubes, and pipe material	
[128]	(a)	Pipe material, length, burial depth, air velocity inside the pipe, and soil types	
[36]	(a)	Burial depth, form factor sigma, and Reynolds number	
[58]	(a)	Pipe length, burial depth, and air velocity	
[59]	(a)	Burial depth, pipe radius, length and ventilation flow rate in the pipe burial depth, pipe radius, pipe length and ventilation flow rate in the tube	
[108]	(a)	Air velocity inside the tube, pipe length, diameter, burial depth, and pipe material	
[28]	(a)	Pipe length	
[72]	(a)	Pipe material, and burial depth	
[121]	(a)	Burial depth, air velocity inside the tubes, and pipe length	
[73]	(b)	Soil types	
[74]	(c)	Ground surface cover	

and it was found that an increase of the spacing between the pipes led to a slight reduction of air temperature at the pipe exit, when the system operated for space cooling.

Different ground surface covers and an environmental parameter (of the third category) were the main factors influencing the EAHE system efficiency, investigated and presented in Ref. [71] for space cooling and [124] for space heating. Two different ground surface boundary conditions were considered: (a) bare soil, and (b) short-grass soil. The cooling potential of the system was investigated and assessed in both cooling and heating operation modes. An extensive sensitivity analysis including the main design parameters of the system (pipe length, radius, burial depth, and air velocity inside the pipe) was presented. Research showed that short grass soil, as ground surface coverage, resulted in a significant improvement of the cooling capacity of the system, while bare soil was found to increase the system's heating capacity.

Multiple pipes were used for cooling and heating a glass-covered 1000 m^2 agricultural greenhouse in Athens (Greece) in Refs. [125, 126] respectively. Sensitivity analyses were performed for cooling and heating operation modes and for the system's design parameters (length and radius of pipes, burial depth, and air velocity inside the pipes). It was shown that for cooling, the greenhouse indoor air temperature decreased with the increase of pipe length and burial depth, while the indoor air temperature increased with the decrease of pipe radius and air velocity inside the tubes. For the heating mode, the system's heating capacity increased with increasing pipe length and burial depth, and decreased with increasing pipe radius and air velocity inside the tubes.

A parametric space cooling investigation was presented in Ref. [109], concerning the influence of two system design parameters: pipe diameter, and air velocity inside the tube. It was found that both parameters had a remarkable effect on the cooling capacity of the system, thus an increase of pipe diameter and air velocity resulted in a significant increase in the air temperature at the pipe exit.

Deglin et al. [60] developed a detailed analysis of the effect of the following system parameters: length, diameter, burial depth, and air velocity inside the tube (first category); and soil type (second category). Two different types of soil were selected: dry sand, and saturated silt; it was found that the most efficient soils were those with higher saturation, thus higher conductivity. As regards the design parameters of the system, the effect of length was significant as greater length provided better results at the pipe exit, while infinite length provides 100% efficiency, with cost a limiting factor. For diameter and air velocity, small pipes were more efficient, while decreasing air velocity resulted in improved efficiency. As regarded burial depth, an increase of depth decreased the influence of the fluctuations of the ground surface temperature, again with cost being a limiting factor.

In [47], a sensitivity analysis was developed regarding the following parameters: pipe diameter, burial depth, and pipe material (first category); and ground surface cover (second category). Heating and cooling efficiency was shown to increase with increasing burial depth, and decreasing with increasing diameter. Plastic, aluminum, concrete, and roughcast were used as pipe materials, and their effect on system efficiency was found to be insignificant. Finally, as regards ground surface coverage, those authors selected a short grass-covered, and an asphalt-covered soil. For a heating operation mode, bare soil was found to increase the efficiency of the system because of its high absorptance.

Lee et al. [57] carried out a parametric investigation of the effect of the following parameters of the first category on the thermal capacity of the system for cooling operation mode: pipe radius, length, air flow rate, and burial depth. Those authors concluded that, for cooling mode, lower air temperature values at the tube exit were achieved with longer pipes, bigger depths, smaller diameters, and lower air velocities inside the tubes. Evidently, cost also played an important role.

The effect of parameters of the first category, such as burial depth, pipe length, air velocity inside tubes, and pipe material on the thermal efficiency of an EAHE system was investigated and analyzed in Ref. [127]. Those authors found that all system parameters could affect

the cooling capacity of the system more than the heating one. A galvanized pipe and a PVC were used, and it was found that outlet air temperature differences were insignificant for both cooling and heating operation modes. An extensive sensitivity analysis was developed and presented in Ref. [128], aimed at investigating and analyzing the EAHE heating and cooling potential for different Italian climates, studying the following parameters of the first category: tube material, tube length, burial depth, and air velocity inside the pipe. It was shown that the effect of pipe material, (PVC, metal, or concrete) on the EAHE performance was negligible. As regards the system design parameters, the preferred choices were 50 m of pipe length, at a burial depth of 3 m, and an air velocity inside the tube equal to 8 m/s.

Sehli et al. [36] carried out a parametric study for cooling, considering burial depth, form factor δ (equal to pipe length over diameter), and the Reynolds number representing the air flow inside the tube. It was shown that an increase of burial depth and form factor resulted in lower air temperature values at the outlet, while an increase of Reynolds number resulted in an increase of the air temperature at the outlet.

A sensitivity analysis was performed in Ref. [58] to study the effect of pipe length, burial depth, and air velocity inside the tube, on the thermal performance of an EAHE system operated in a closed loop mode, and used for space cooling in hot and humid climates. Pipe length, burial depth, and air velocity were found to play an important role on the system efficiency, as the cooling capacity of EAHE increased with longer tubes, increased depth, and smaller velocities. Those authors proposed to take these parameters into account simultaneously in order to optimize the performance of the system.

In an extensive parametric study, Yang et al. [59] analyzed the influence of the following (first category) system design parameters on the air temperature fluctuation at the EAHE exit: burial depth, pipe radius, pipe length, and ventilation flow rate in the tube. It was confirmed that both pipe length and burial depth are the most significant systems' parameters. In particular, the increase of both pipe length and burial depth result in a remarkable improvement of the system's cooling capacity. An increase of the pipe radius and the air flow rate resulted in a slight increase of outlet air temperature.

Ahmed et al. [108] offered an extensive sensitivity analysis related to the thermal performance of a horizontal EAHE system for space cooling. The considered (first category) parameters were: air velocity inside the tube; pipe length, diameter, and material; and burial depth. As regards air velocity, four values were considered: 0.41, 1.0, 1.5, and 2.0 m/s; it was found that 1.5 m/s offered the best cooling capacity. Pipe lengths of 7.5, 15, 30, and 60 m were considered, and the 60 m gave the optimum performance with velocity equal to 1.5 m/s. The smallest pipe diameter offered the most improved performance. As regards pipe material, PVC, polyethylene, concrete, and clay were used, and it was found that the material with the highest conductivity gave the most improved cooling performance. Finally, burial depths of 0.6, 2.0, 4.0, and 8 m were used in simulations, and cooling effect was found to increase with depth.

In an EAHE study, Belatrache et al. [28] investigated the effect of pipe length on the air temperature difference between inlet and outlet air during the summer period. The air temperature difference between the inlet and outlet of a pipe was found to increase significantly with length up to about 25 m. For bigger lengths, the temperature difference became insignificant. In Ref. [72], a study of the influence of two (first category) parameters, pipe material and burial depth, on the thermal efficiency of an EAHE system was undertaken. Two pipe materials were considered, steel and PVC, and it was shown that the material selection did not offer a significant effect on system efficiency. For burial depth, those authors concluded that a depth of 100 cm would be an optimal choice for an improved system performance.

Wei et al. [121] carried out an extensive experimental investigation of the cooling capacity of an EAHE system for hot and humid climates. Those authors also carried out a sensitivity analysis using the following (first category) system parameters: burial depth, air velocity inside the tubes, and pipe length. An increase of burial depth resulted in improving drastically the air temperature at the outlet, and the cooling capacity of the system. Decreasing the pipe diameter resulted in minimizing the outlet air temperature and humidity, but it did not improve the cooling capacity of the system. An increase of air velocity resulted in an increase of outlet temperature for the same length, while the air temperature at the pipe outlet decreased exponentially with increasing pipe length.

An investigation of the influence of soil thermal conductivity on the thermal behavior of an EAHE system for space cooling was done in Ref. [73]. Soil thermal conductivities equal to 0.52, 2, and $4WK^{-1}m^{-1}$, resulted in a maximum air temperature drop of 15.6, 17.0 and 17.3 K. The type of soil coating, which was tested in Ref. [74] to calculate its effect on the thermal performance of EAHE for cooling and heating, was found to play a significant role on the thermal performance of a system. Three types of soil coating, sand, sand-bent (a mix of sand and bentonite), and in-situ earth, were combined with two types of moisture (minimum and maximum). The system thermal performance presented a difference of up to 15.9% for the same soil coating with the two different moisture contents, and a difference of up to 17.4% for minimum moisture with the different soil coating types. For the maximum moisture content, the soil coating type resulted in an impact of only 2% on the thermal performance. A summary of the aforementioned parametric studies is shown in Table 4.

5. Hybrid EAHE-renewable systems

In all EAHE systems, a small amount of electricity is necessary to transfer both thermal and cooling load to specific locations within buildings and structures. This is also necessary to meet the needs of air conditioning. The required electrical energy varies according to the type of size applications. In addition, the possibility of coupling EAHE systems with energy systems producing heat is a particularly effective solution to reach maximum heat production with maximum efficiency.

In the previous sections of this work, a number of the discussed types of EHAE systems referred to their autonomous operation, using conventional power sources. In this section, recent research papers that confirm the fact that the effectiveness and utility of an EAHE system may be increased by coupling it with active Renewable Energy Systems (RES), are discussed. This way, a hybrid system is defined, which consists of a specific type of EAHE in combination with several renewable

Table 4

Summary of the results of parametric studies.

PARAMETRIC STUDIES RESULTS			
System's Parameter	Reference	Results	
Pipe length	[28,36,43,57–60, 70,71,108, 121–128]	An increase resulted in a significant improvement of system efficiency	
Pipe radius/ diameter	[36,43,47,57,59,60, 70,71,108,109, 122–126]	An increase resulted in a reduction of system efficiency	
Air velocity inside the tube/air mass rate	[43,57–60,70,71, 108,109,121–128]	An increase resulted in a reduction of system efficiency	
Burial depth	[36,43,47,57–60, 71,72,108, 121–128]	An increase resulted in a significant improvement of system efficiency	
Pipe material	[47,72,108,127, 128]	No significant impact on the system efficiency	
Space between adjacent pipes	[43]	An increase resulted in an improvement of system efficiency	
Ground surface cover	[47,71,74,124]	Short-grass covered soil improved the system's cooling capacity while bare soil improved the heating one. Soil surface cover played an important role in the EAHE system thermal performance.	
Soil types	[60,73,128]	Higher conductivity soils were the most efficient	

energy (RE) technologies so that the requirements of thermal loads may be addressed, something that a standalone EAHE would be unable to do.

In a study published in 2019, Agrawal et al. [2] discussed the state-of-the-art for applications, technology integration, and the latest research trends on earth-air-heat exchanger systems. Among the studied systems, there were recent works of importance, including hybrid EAHE systems coupled with solar energy systems. Elminshawy et al. [90] used EAHE system for cooling solar (photovoltaic, PV) panels. In particular, they used an EHAE system that cooled a hybrid photovoltaic thermal (PVT) system with air heat exchanger on its rear side, as depicted in Fig. 10. That study revealed that, by passing the EAHE cooled air to a solar PV, the temperature of the solar PV module decreased by almost 24% and the electrical conversion efficiency and output power of the PV module increased by 23% and 19%, respectively.

Following the concept of the operational combination of EAHE and PVT systems, Jakhar et al. [91] studied the effectiveness of an earth-water heat exchanger system in cooling the PV panels. In that study, the maximum PV panel temperature was almost 75 °C without cooling, while it decreased remarkably down to almost 46 °C with EHAE cooling, for low values of water flow rate.

Most research workers rarely touch upon the improvement of performance of air-cooled PVT systems employing an active geothermal air cooling using a buried EAHE in closed cycle with the (already existing) standalone PV. Yildiz et al. [92] investigates a hybrid EAHE system coupled with PV cells, focusing on the overall performance improvement. That work analyzed the PV assisted closed-loop EAHE exchanger for cooling a greenhouse. The results indicated that 31% of electric energy could be obtained from solar PV cells and the rest 69% of obtained from conventional resources, resulting in total energy savings of almost 30%.

Zapałowicz and Opiela [93] proposed a hybrid air-conditioning system for a building. An EAHE and a PV unit gave a more sophisticated solution. The ambient air was first cooled through the EAHE system, and some of this cooled air flew through a channel formed by the rear side of the PV modules and the wall of the building, reducing the temperature of the PV modules and improving their efficiency. At the same time, the building wall also cooled down and reduced the heat gain into the building. The remaining cooled air was directed to the air conditioner to lower the air conditioning load, as shown in Fig. 11.

Hachchadi et al. [76] presented a similar performance analysis, which described a PV thermal air collector combined with water-to-air heat exchanger for renewed air conditioning in buildings. The operational of the hybrid system was improved, minimizing the required area of the heat exchanger by about 33%.

On the use of PV or hybrid PVT units in EAHE systems, recent efforts at optimization have been conducted, looking for the optimum values of all parameters. Li et al. [77] accomplished a multi-objective energy and exergy optimization analysis for different configurations of hybrid earth-air heat exchanger coupled with a Building Integrated Photovoltaic/thermal system (BIPVT). The multi-criteria decision making procedure was performed to maximize the yearly total energy and exergy outputs of different configurations. The length, depth, and width of the channel located under the PV modules, air mass flow rate, and the length and diameter of the EAHE system were the decision variables.

Active solar thermal combined with EAHE systems, have proven to be the most promising solution for the effective operation of several types of heat exchangers. Many researchers have examined this coupling over the last decade. Wang et al. [78] studied experimentally an underground thermal storage in a solar-ground-coupled heat pump system for residential buildings. That study implied that the performance of underground thermal storage of a hybrid EAHE is strongly dependent on the intensity of solar radiation, and the matching between the water tank volume and the area of solar collectors. The overall efficiency of the hybrid system improved over 20%, while the efficiency of the absorbed solar energy exceeded 70%. It is obvious that such hybrid EAHE systems operate more effectively than conventional ones.

A multi-objective optimization of a seasonal solar thermal energy storage system combined with an EAHE exchanger and a solar collector field was researched in the work of Benzaana et al. [79], where the produced thermal energy goes to net zero energy buildings (NZEBs). Fig. 12 depicts the proposed installation conceptually, considering the basic parameters. Experimentally validated numerical results showed that the combination of the hybrid EAHE system satisfied all energy needs, covering 45% of the total thermal energy all year long.

A similar approach was attempted earlier by Attar et al. [80], who used a combination of parametric and numerical study of a solar system for heating a greenhouse equipped with a buried exchanger. In that study, a TRNSYS simulation project was designed, following a determination process for the experimental input parameters in the TRNSYS environment. The results were experimentally validated and showed that the volume of the storage tank in connection with the absorbing area of the solar collector field are the most significant parameters. In particular, the lower flow rate in the heat exchanger and the minimization of the temperature stratification of the storage tank could increase the air temperature inside the greenhouse by 5 $^{\circ}$ C.

Passive cooling of buildings through geothermal systems has been one of the most important applications of EAHE system. Their effectiveness was significantly improved by the use of solar air heating ducts and solar chimneys. Jakhar et al. [88] presented research on the thermal performance of Earth Air Tunnel Heat Exchanger (EATHE) coupled with a solar air-heating duct. It was shown that the heating capacity of EATHE system increased by over 5% when it was coupled with solar air heating duct, with a substantial increase in the room temperature by an average of 2.3 °C higher than the base case.

Maerefat and Haghighi [81]introduced the use of a solar chimney together with EAHE systems. The conducted theoretical analysis investigated the cooling and ventilation in a solar house through combined



Fig. 10. Schematic diagram and experimental setup of hybrid EAHE coupled with PVT collector [90].



Fig. 11. Schematic diagram of proposed hybrid EAHE installation [93].



Fig. 12. EAHE system coupled with underground tank and solar collector for heating [79].

solar chimney and an underground air channel. The findings showed that the solar chimney powered the underground cooling system during daytime, without any need to consume electricity. Although the temperature and cooling load were high, the hybrid system could provide satisfactory thermal comfort conditions, even if the solar density were low, less than 100 W/m², and the temperature were extremely high, over 50 °C. In a later work [82], those same authors provided a design guideline for the application of EAHE systems coupled with solar chimney as a natural heating system, in order to meet the thermal need of flat buildings in respect to adaptive thermal comfort criteria. The proposed model was validated and showed that the overall performance of such hybrid EAHE systems depends strongly on solar radiation and air temperature. For any values of these two parameters, the optimum size of a solar chimney is 0.2 m for the air gap, and 0.5 m for the diameter of the heating pipe.

In an interesting work, Yu et al. [89] studied experimentally the coupling of a geothermal cooling system with earth tube and solar chimney. Room air quality, thermal comfort, thermal capabilities, and underground soil temperature analyses were performed for 43 days. The results showed that the coupled geothermal system could provide cooling to the facility in a natural operation mode without using any electricity, while the surrounding soil of the earth tube could saturate

and take long to recover if the heat is over-extracted from the soil in a forced air mode. Sakhri et al. [83] have suggested an effective idea of integrating a solar chimney into a geothermal system, with the main characteristic being the east-west orientation of the facility. The air inlet would be placed in the east and be heat-insulated, in contrast to the outlet, which would be placed in the west. The air temperature difference between outlet and inlet is capable of reducing energy consumption significantly, and ultimately improving the overall performance of the hybrid system.

An alternative look at the subject of hybrid EAHE systems was provided by Benhammou et al. [84], where they proposed a typical EAHE system (buried pipe in PVC), coupled with a wind tower. A transient analytical model was developed to investigate the influence of design parameters on the performance of the EAHE. The model of the EAHE was validated against both theoretical and experimental data. The results showed that the influence of the EAHE dimensions again dominated, while the mean efficiency and the gradient of air temperature increased with (increasing) pipe length, but decreased with (increasing) pipe diameter.

The most comprehensive hybrid EAHE system proposal consisting of an electric boiler, hydrogen, wind, and PV configurations, was provided by Akhtari et al. [75]. The heat exchanger achieved improved reliability and sustainability by coupling with a hybrid renewable energy system, including wind, solar and hydrogen. The results showed that adding a hybrid RE system to an EAHE system could lead to an improvement of about 5.5% of the renewable fraction, and decreased emissions and diesel consumption by almost 48%.

6. Economic assessment of earth to air heat exchangers

Much of residential energy (consumed in households) is customarily used for heating, cooling, air conditioning, and production of hot water. Thus, residential energy consumption has a significant share in global energy consumption [3]. Systems that utilize RES can be a solution for the creation of buildings that are sustainable and friendly to the environment. As one such application, the ground source heat pump system is an alternative solution that can help reduce residential energy consumption [19].

The earth to air heat exchanger (EAHE) is a passive technique that is based on the usage of the underground soil temperature [18]. It consists of one or more pipes, made from metal, plastic or concrete, which are laid underground horizontally [11,28]. EAHE systems do not require high maintenance and have a high potential for energy saving [74]. They are suitable for the heating or cooling of small and medium spaces, and have low initial and operational costs [129].

The evaluation of the performance of earth to air heat exchangers and their economic assessment is particularly interesting and has been the subject of recent literature. For instance, Chel and Tiwari [130] assessed the performance of such a system in a New Delhi (India) building and provided a life cycle cost analysis, concluding that the integration of earth to air heat exchangers has significant potential for energy saving, estimated at 10,321 kWh/year, compared to 4,946 kWh/year that was the energy saving potential before the integration. The annual cash inflow was estimated at \$448/year and the annual cash outflow (operation and maintenance) at \$240.9/year.

Ascione et al. [128] performed a technoeconomic analysis and evaluated the performance of an earth to air heat exchanger for both winter and summer in different Italian climates. It was found that such a system presented the highest efficiency for winter and summer in colder climates. In addition, if earthworks may be done inexpensively and easily, the system could be economically acceptable with simple payback estimated at 5–9 years.

Akhtari et al. [75] also performed a technoeconomic evaluation for an earth to air heat exchanger, coupled with a hybrid energy system including wind energy, solar energy and hydrogen. It was found that effectiveness rose by 8% during an intermittent run of the system, and energy supply was around 31.55 MJ per month. If geothermal energy was added to the hybrid system, the renewable fraction could be improved by approximately 5.5%, while emissions and diesel consumption were reduced by approximately 48%.

Bisoniya et al. [131] assessed the performance of an EAHE system in the hot and dry conditions of Bhopal (India). The embodied energy of the system was estimated at 1663.88 kWh, while its maximum heating and cooling potential were found to be 191.06 kWh in January and 247.25 kWh in May. The energy payback period was calculated to be 1.29 years and, in its 50 years lifespan, the earned carbon credit equaled \$2837.6, while the CO_2 emission mitigation potential was estimated at 101.3 tons.

Li et al. [132] focused on the evaluation of a passive system that coupled EAHE with solar chimney. Such a system could provide up to 2, 582 W of cooling capacity, covering the cooling load of the building and maintaining thermal comfort. In another work, Li et al. [133] evaluated an EAHE system preheating fresh air in cold climate, finding that the EAHE system could lead to a temperature increase that could reach 12.4 °C with an average heating coefficient of performance estimated at 29.7, meaning that system was an effective and economic technique.

Fazlikhani et al. [27] studied the efficiency of an EAHE system in both hot and cold climates in Iran. The system could lead to improvements of the average temperature and to significant reductions in energy consumption. It was suggested that the system could be used for 294 days per year to provide energy savings of 50.1–63.6% in hot climates; and for 225 days per year, providing energy savings of 24.5–47.9% in cold climates.

Díaz-Hernández et al. [134] examined the case of an EAHE in Mexico and evaluated the system's performance under a warm humid climate. The system worked as a cooler in the morning and as a heater at night, while in the winter it could work as a heater for most of the day. Those authors also performed an analysis based on an experimental 6-month period, when the EAHE was coupled with an air conditioning system. It was estimated that the payback period was between 1 and 2 years, with the EAHE cost being \$132.2 and the percentages of energy savings being 20–21% for most months.

Nemati et al. [135]performed an evaluation of the performance of a hybrid cooling system that combined an earth to air heat exchanger with an indirect evaporative cooler. This combination improved cooling performance, assisted in maintaining thermal comfort, and reduced energy and water consumption by approximately 62 and 45%, correspondingly. Various economic parameters were estimated, including initial investment, initial cost-saving, and annual cost savings (reduced maintenance costs, savings due to reduced electricity and natural gas demand, reduced carbon emission tax), and it was concluded that the payback time was between 5.5 and 6.4 years.

Ahmadi et al. [136] evaluated the performance of a hybrid cooling system, including an EAHE and a water spray channel, in Tehran (Iran). That work suggested that its effectiveness exceeded 100%, indicating that such a system could be used as an alternative to conventional coolers since it could provide optimal thermal comfort in the summer. Such hybrid systems are both ecofriendly and energy efficient. It was estimated that the operation cost saving could be up to 71%, with a payback period of 286 days.

7. Conclusions

Passive heating/cooling applications and technologies have gained ground during the last decades in order to achieve energy conservation in buildings and improve indoor and outdoor thermal comfort conditions and microclimate. Among these, earth-to-air heat exchangers have become a remarkably attractive subject for research and implementation. The present review article presented a holistic approach to the EAHE systems, including modeling the ground temperature distribution and the thermal performance of the EAHE systems; experimental evaluation; parametric studies; novel advances of hybrid technologies; and an economic assessment of system implementation. Concluding remarks are summarized below.

The ground temperature distribution at the surface and various depths below it, depend on the energy balance equation at the surface, containing the following crucial parameters: (a) solar radiation; (b) terrestrial and atmospheric radiation (long-wave); (c) convective energy between air and ground; and (d) latent heat flux due to evaporation/ condensation processes. The ground temperature variation follows a sinusoidal distribution with an annual fluctuation up to 4 °C and a daily one up to 0.3 °C.

Many models predicting the thermal performance of EAHE systems have been developed and presented in the scientific literature. These models could be divided in three categories: (a) numerical (offering numerical solutions of complicated thermal processes including heat conduction in the soil, and convection in the soil, the tube and the air); (b) analytical (solving the heat and mass transfer equations analytically); and (c) data-driven (where modeling and prediction is achieved by training and testing the designed network with historical data). Most of the presented models were experimentally validated and had successfully predicted the thermal behavior of the systems.

A presentation of parametric studies followed, proving the importance of the main configuration parameters on system efficiency. System parameters were divided into three categories: (a) system design (pipe length and radius, burial depth, air velocity inside the tube, number of pipes, and pipe material); (b) soil types; and (c) soil surface coverage. System design parameters, especially length and burial depth, were shown to bear the most important influence on the thermal efficiency of the system.

Conducted experimental work has been mostly empirical. The reported findings are important and sufficient to highlight potential future research. As computer power and electronics evolve with unprecedented velocity and the Internet allows research findings to become rapidly accessible to the scientific community, more scientists focus on modeling. This has become obvious from the present review. However, a model requires assumptions, equations, and validation. This review has demonstrated that experimental research, which plays a key role in model validation but also sets the base for the assumptions and the range of model applicability, is mostly empirical. To support the effort of researchers working on quality model development, experimental work needs to be expanded and enhanced, so that models can drive a faster and more reliable application of heating and cooling by means of buried pipes.

While the measurement of global properties and the overall performance of a system are necessary, there are enough studies to pinpoint what the next step should be. The role of heat accumulation is an important parameter which requires more attention. Thermophysical properties of a great range of materials are poorly known and in most such applications effective properties are used. Engineering designs employing low accuracy data will inevitably provide results with a large error [137]. It is important to design appropriate laboratory experimental simulators to study the effects of individual factors or appropriate groups. Synergistic effects of natural factors should be considered, studied, and implemented. Traditional residencies and ancient building choices may provide insight as to how synergistic effects can dramatically increase the system efficiency. Finally, looking at this issue holistically, the reduction of energy consumption in a small system, such as a building, should not be examined in isolation from its environment. The energy consumption and potential pollution or ecological consequences arising from the construction and operation of such systems have to be foreseen, identified, evaluated in terms other than economic and minimalized.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

The authors thank Drs. T. Nadasdi and S. Sinclair for their online Spell Check Plus (https://spellcheckplus.com) that was used for proofing the entire manuscript. The authors also thank all journals that gave us permission for using Figs. 1a to 12 (with the following order numbers from https://s100.copyright.com/: 5105800682691, 5105800944042, 5105801157401, 5105801372321, 5105810042286, 5105810189508, 5105810314649, 5105810540714, 5154640682921, 5154641398083, 5154650456771).

References

- Singh R, Sawhney RL, Lazarus IJ, Kishore VVN. Recent advancements in earth air tunnel heat exchanger (EATHE) system for indoor thermal comfort application: a review. Renew Sustain Energy Rev 2018;82:2162–85.
- [2] Agrawal KK, Misra R, Das Agrawal G, Bhardwaj M, Jamuwa DK. The state of art on the applications, technology integration, and latest research trends of earthair-heat exchanger system. Geothermics 2019;82:34–50.
- [3] Karytsas S, Theodoropoulou H. Public awareness and willingness to adopt ground source heat pumps for domestic heating and cooling. Renew Sustain Energy Rev 2014;34:49–57.

- [4] Mustafa Omer A. Ground-source heat pumps systems and applications. Renew Sustain Energy Rev 2008;12(2):344–71.
- [5] Omer AM. Energy, environment and sustainable development. Renew Sustain Energy Rev 2008;12(9):2265–300.
- [6] Lund H. Renewable energy strategies for sustainable development. Energy 2007; 32(6):912–9.
- [7] Santamouris M, Kolokotsa D. Passive cooling dissipation techniques for buildings and other structures: the state of the art. Energy Build 2013;57:74–94.
- [8] Santamouris M, Pavlou K, Synnefa A, Niachou K, Kolokotsa D. Recent progress on passive cooling techniques. Advanced technological developments to improve survivability levels in low-income households. Energy Build 2007;39(7):859–66.
- [9] Santamouris M, et al. Using cool paving materials to improve microclimate of urban areas - design realization and results of the flisvos project. Build Environ 2012;53:128–36.
- [10] Santamouris M, Synnefa A, Karlessi T. Using advanced cool materials in the urban built environment to mitigate heat islands and improve thermal comfort conditions. Sol Energy 2011;85(12):3085–102.
- [11] Bisoniya TS, Kumar A, Baredar P. Experimental and analytical studies of earth-air heat exchanger (EAHE) systems in India: a review. Renew Sustain Energy Rev 2013;19:238–46.
- [12] Goswami DS, Ileslamlou S. Performance analysis of a closed-loop climate control system using underground air tunnel. J. Sol. Energy Eng. Trans. ASME 1990;112 (2):76–81.
- [13] Givoni B, Katz L. Earth temperatures and underground buildings. Energy Build 1985;8(1):15–25.
- [14] Jayashankar BC, Sawhney RL, Sodha MS. Effect of different surface treatments of the surrounding earth on thermal performance of earth-integrated buildings. Int J Energy Res 1989;13(5):605–19.
- [15] Mihalakakou G, Santamouris M, Asimakopoulos D. Modelling the earth temperature using multiyear measurements. Energy Build 1992;19(1):1–9.
- [16] Vaz J, Sattler MA, Dos Santos ED, Isoldi LA. Experimental and numerical analysis of an earth-air heat exchanger. Energy Build 2011;43(9):2476–82.
- [17] Sawhney RL, Buddhi D, Thanu NM. An experimental study of summer performance of a recirculation type underground airpipe air conditioning system. Build Environ 1998;34(2):189–96.
- [18] Ozgener L. A review on the experimental and analytical analysis of earth to air heat exchanger (EAHE) systems in Turkey. Renew Sustain Energy Rev 2011;15 (9):4483–90.
- [19] Peretti C, Zarrella A, De Carli M, Zecchin R. The design and environmental evaluation of earth-to-air heat exchangers (EAHE). A literature review. Renew Sustain Energy Rev 2013;28:107–16.
- [20] Kaushal M. Geothermal cooling/heating using ground heat exchanger for various experimental and analytical studies: comprehensive review. Energy Build 2017; 139:634–52.
- [21] Tittelein P, Achard G, Wurtz E. Modelling earth-to-air heat exchanger behaviour with the convolutive response factors method. Appl Energy 2009;86(9):1683–91.
- [22] Darkwa J, Kokogiannakis G, Magadzire CL, Yuan K. Theoretical and practical evaluation of an earth-tube (E-tube) ventilation system. Energy Build 2011;43 (2–3):728–36.
- [23] Trzaski A, Zawada B. The influence of environmental and geometrical factors on air-ground tube heat exchanger energy efficiency. Build Environ 2011;46(7): 1436–44.
- [24] Gan G. Simulation of dynamic interactions of the earth-air heat exchanger with soil and atmosphere for preheating of ventilation air. Appl Energy 2015;158: 118–32.
- [25] Xamán J, Hernández-López I, Alvarado-Juárez R, Hernández-Pérez I, Álvarez G, Chávez Y. Pseudo transient numerical study of an earth-To-Air heat exchanger for different climates of México. Energy Build 2015;99:273–83.
- [26] Niu F, Yu Y, Yu D, Li H. Investigation on soil thermal saturation and recovery of an earth to air heat exchanger under different operation strategies. Appl Therm Eng 2015;77:90–100.
- [27] Fazlikhani F, Goudarzi H, Solgi E. Numerical analysis of the efficiency of earth to air heat exchange systems in cold and hot-arid climates. Energy Convers Manag 2017;148:78–89.
- [28] Belatrache D, Bentouba S, Bourouis M. Numerical analysis of earth air heat exchangers at operating conditions in arid climates. Int J Hydrogen Energy 2017; 42(13):8898–904.
- [29] Rodrigues MK, et al. Thermal performance simulations of Earth-Air Heat Exchangers for different soils of a coastal city using in-situ data. Sustain Energy Technol Assess 2018;30:224–9.
- [30] Liu Z, et al. Numerical modeling and parametric study of a vertical earth-to-air heat exchanger system. Energy 2019;172:220–31.
- [31] Hollmuller P, Lachal B. Cooling and preheating with buried pipe systems:
- monitoring, simulation and economic aspects. Energy Build 2001;33(5):509–18.[32] Puri VM. Heat and mass transfer analysis and modeling in unsaturated ground soils for buried tube systems. Energy Agric 1987;6(3):179–93.
- [33] Wu H, Wang S, Zhu D. Modelling and evaluation of cooling capacity of earth-airpipe systems. Energy Convers Manag 2007;48(5):1462–71.
- [34] Zhang J, Haghighat F. Convective heat transfer prediction in large rectangular cross-sectional area Earth-to-Air Heat Exchangers. Build Environ 2009;44(9): 1892–8.
- [35] Su H, Liu XB, Ji L, Mu JY. A numerical model of a deeply buried air-earth-tunnel heat exchanger. Energy Build 2012;48:233–9.
- [36] Sehli A, Hasni A, Tamali M. The potential of earth-air heat exchangers for low energy cooling of buildings in South Algeria," in. Energy Procedia 2012;18: 496–506.

G. Mihalakakou et al.

- [37] Brum RDS, Vaz J, Rocha LAO, Dos Santos ED, Isoldi LA. A new computational modeling to predict the behavior of Earth-Air Heat Exchangers. Energy Build 2013;64:395–402.
- [38] Misra R, Bansal V, Das Agrawal G, Mathur J, Aseri T. Transient analysis based determination of derating factor for Earth Air Tunnel Heat Exchanger in winter. Energy Build 2013;58:76–85.
- [39] Ramírez-Dávila L, Xamán J, Arce J, Álvarez G, Hernández-Pérez I. Numerical study of earth-to-air heat exchanger for three different climates. Energy Build 2014;76:238–48.
- [40] Mathur A, Surana AK, Mathur S. Numerical investigation of the performance and soil temperature recovery of an EATHE system under intermittent operations. Renew Energy 2016;95:510–21.
- [41] Serageldin AA, Abdelrahman AK, Ookawara S. Earth-Air Heat Exchanger thermal performance in Egyptian conditions: experimental results, mathematical model, and Computational Fluid Dynamics simulation. Energy Convers Manag 2016;122: 25–38.
- [42] Mihalakakou G, Santamouris M, Asimakopoulos D. Modelling the thermal performance of earth-to-air heat exchangers. Sol Energy 1994;53(3):301–5.
- [43] Mihalakakou G, Santamouris M, Asimakopoulos D. Use of the ground for heat dissipation. Energy 1994;19(1):17–25.
- [44] Bojic M, Trifunovic N, Papadakis G, Kyritsis S. Numerical simulation, technical and economic evaluation of air-to-earth heat exchanger coupled to a building. Energy 1997;22(12):1151–8.
- [45] Gauthier C, Lacroix M, Bernier H. Numerical simulation of soil heat exchangerstorage systems for greenhouses. Sol Energy 1997;60(6):333–46.
- [46] Kabashnikov VP, Danilevskii LN, Nekrasov VP, Vityaz IP. Analytical and numerical investigation of the characteristics of a soil heat exchanger for ventilation systems. Int J Heat Mass Tran 2002;45(11):2407–18.
- [47] Badescu V. Simple and accurate model for the ground heat exchanger of a passive house. Renew Energy 2007;32(5):845–55.
- [48] Yoon G, Tanaka H, Okumiya M. Study on the design procedure for a multi-cool/ heat tube system. Sol Energy 2009;83(8):1415–24.
- [49] Sodha MS, Sharma AK, Singh SP, Bansal NK, Kumar A. Evaluation of an earth-air tunnel system for cooling/heating of a hospital complex. Build Environ 1985;20 (2):115–22.
- [50] Santamouris M, Lefas CC. Thermal analysis and computer control of hybrid greenhouses with subsurface heat storage. Energy Agric 1986;5(2):161–73.
- [51] Rouag A, Benchabane A, Mehdid CE. Thermal design of Earth-to-Air Heat Exchanger. Part I a new transient semi-analytical model for determining soil temperature. J Clean Prod 2018;182:538–44.
- [52] Mihalakakou G, Santamouris M, Asimakopoulos D, Tselepidaki I. Parametric prediction of the buried pipes cooling potential for passive cooling applications. Sol Energy 1995;55(3):163–73.
- [53] Krarti M, Lopez-Alonzo C, Claridge DE, Kreider JF. Analytical model to predict annual soil surface temperature variation. J. Sol. Energy Eng. Trans. ASME 1995; 117(2):91–9.
- [54] Bojić M, Papadakis G, Kyritsis S. Energy from a two-pipe, earth-to-air heat exchanger. Energy 1999;24(6):519–23.
- [55] Al-Ajmi F, Loveday DL, Hanby VI. The cooling potential of earth-air heat exchangers for domestic buildings in a desert climate. Build Environ 2006;41(3): 235–44.
- [56] Cucumo M, Cucumo S, Montoro L, Vulcano A. A one-dimensional transient analytical model for earth-to-air heat exchangers, taking into account condensation phenomena and thermal perturbation from the upper free surface as well as around the buried pipes. Int J Heat Mass Tran 2008;51(3–4):506–16.
- [57] Lee KH, Strand RK. The cooling and heating potential of an earth tube system in buildings. Energy Build 2008;40(4):486–94.
- [58] Do SL, Baltazar JC, Haberl J. Potential cooling savings from a ground-coupled return-air duct system for residential buildings in hot and humid climates. Energy Build 2015;103:206–15.
- [59] Yang D, Guo Y, Zhang J. Evaluation of the thermal performance of an earth-to-air heat exchanger (EAHE) in a harmonic thermal environment. Energy Convers Manag 2016;109:184–94.
- [60] Deglin D, Van Caenegem L, Dehon P. Subsoil heat exchangers for the air conditioning of livestock buildings. J Agric Eng Res 1999;73(2):179–88.[61] Mihalakakou G. On the heating potential of a single buried pipe using
- deterministic and intelligent techniques. Renew Energy 2003;28(6):917–27. [62] Kumar R, Kaushik SC, Garg SN. Heating and cooling potential of an earth-to-air
- heat exchanger using artificial neural network. Renew Energy 2006;31(8): 1139–55.
 [63] Diaz SE, Sierra JMT, Herrera JA. The use of earth-air heat exchanger and fuzzy
- [63] Diaz SE, Sierra JMI, Herrera JA. The use of earth-air near exchanger and fuzzy logic control can reduce energy consumption and environmental concerns even more. Energy Build 2013;65:458–63.
- [64] Pfafferott J. Evaluation of earth-to-air heat exchangers with a standardised method to calculate energy efficiency. Energy Build 2003;35(10):971–83.
- [65] Ghosal MK, Tiwari GN, Srivastava NSL. Thermal modeling of a greenhouse with an integrated earth to air heat exchanger: an experimental validation. Energy Build 2004;36(3):219–27.
- [66] Tiwari GN, Akhtar MA, Shukla A, Emran Khan M. Annual thermal performance of greenhouse with an earth-air heat exchanger: an experimental validation. Renew Energy 2006;31(15):2432–46.
- [67] Chiesa G, Simonetti M, Grosso M. A 3-field earth-heat-exchange system for a school building in Imola, Italy: monitoring results. Renew Energy 2014;62: 563–70.
- [68] Khabbaz M, Benhamou B, Limam K, Hollmuller P, Hamdi H, Bennouna A. Experimental and numerical study of an earth-to-air heat exchanger for air

Renewable and Sustainable Energy Reviews 155 (2022) 111921

cooling in a residential building in hot semi-arid climate. Energy Build 2016;125: 109–21.

- [69] Liu Z, et al. Experimental investigation of a vertical earth-to-air heat exchanger system. Energy Convers Manag 2019;183:241–51.
- [70] Sodha MS, Buddhi D, Sawhney RL. Optimization of pipe parameters of an underground air pipe cooling system. Energy Convers Manag 1993;34(6):465–70.
- [71] Mihalakakou G, Santamouris M, Asimakopoulos D, Papanikolaou N. Impact of ground cover on the efficiencies of earth-to-air heat exchangers. Appl Energy 1994;48(1):19–32.
- [72] Sakhri N, Menni Y, Ameur H. Effect of the pipe material and burying depth on the thermal efficiency of earth-to-air heat exchangers. Case Stud Chem Environ Eng 2020;2:100013.
- [73] Bansal V, Misra R, Das Agarwal G, Mathur J. Transient effect of soil thermal conductivity and duration of operation on performance of Earth Air Tunnel Heat Exchanger. Appl Energy 2013;103:1–11.
- [74] Cuny M, Lin J, Siroux M, Magnenet V, Fond C. Influence of coating soil types on the energy of earth-air heat exchanger. Energy Build 2018;158:1000–12.
- [75] Akhtari MR, Shayegh I, Karimi N. Techno-economic assessment and optimization of a hybrid renewable earth - air heat exchanger coupled with electric boiler, hydrogen, wind and PV configurations. Renew Energy 2020;148:839–51.
- [76] O. Hachchadi, M. Bououd, and A. Mechaqrane, "Performance analysis of photovoltaic-thermal air collectors combined with a water to air heat exchanger for renewed air conditioning in building," Environ Sci Pollut Res, vol. 28, no. 15, pp. 18953–18962, 2021.
- [77] Li ZX, Shahsavar A, Al-Rashed AAAA, Kalbasi R, Afrand M, Talebizadehsardari P. Multi-objective energy and exergy optimization of different configurations of hybrid earth-air heat exchanger and building integrated photovoltaic/thermal system. Energy Convers Manag 2019;195:1098–110.
- [78] Wang H, Qi C, Wang E, Zhao J. A case study of underground thermal storage in a solar-ground coupled heat pump system for residential buildings. Renew Energy 2009;34(1):307–14.
- [79] Benzaama MH, Menhoudj S, Lekhal MC, Mokhtari A, Attia S. Multi-objective optimisation of a seasonal solar thermal energy storage system combined with an earth – air heat exchanger for net zero energy building. Sol Energy 2021;220: 901–13.
- [80] Attar I, Naili N, Khalifa N, Hazami M, Farhat A. Parametric and numerical study of a solar system for heating a greenhouse equipped with a buried exchanger. Energy Convers Manag 2013;70:163–73.
- [81] Maerefat M, Haghighi AP. Passive cooling of buildings by using integrated earth to air heat exchanger and solar chimney. Renew Energy 2010;35(10):2316–24.
- [82] Haghighi AP, Maerefat M. Design guideline for application of earth-toair heat exchanger coupled with solar chimney as a natural heating system. Int J Low-Carbon Technol 2013;10(3):294–304.
- [83] Sakhri N, et al. Experimental study of an earth-to-air heat exchanger coupled to the solar chimney for heating and cooling applications in arid regions. J Therm Anal Calorim 2021;145:3349–58.
- [84] Benhammou M, Draoui B, Zerrouki M, Marif Y. Performance analysis of an earthto-air heat exchanger assisted by a wind tower for passive cooling of buildings in arid and hot climate. Energy Convers Manag 2015;91:1–11.
- [85] Zhou T, Xiao Y, Liu Y, Lin J, Huang H. Research on cooling performance of phase change material-filled earth-air heat exchanger. Energy Convers Manag 2018; 177:210–23.
- [86] Soni SK, Pandey M, Bartaria VN. Energy metrics of a hybrid earth air heat exchanger system for summer cooling requirements. Energy Build 2016;129:1–8.
- [87] Go GH, Lee SR, Yoon S, Kim MJ. Optimum design of horizontal ground-coupled heat pump systems using spiral-coil-loop heat exchangers. Appl Energy 2016;162: 330–45.
- [88] Jakhar S, Misra R, Bansal V, Soni MS. Thermal performance investigation of earth air tunnel heat exchanger coupled with a solar air heating duct for northwestern India. Energy Build 2015;87:360–9.
- [89] Yu Y, Li H, Niu F, Yu D. Investigation of a coupled geothermal cooling system with earth tube and solar chimney. Appl Energy 2014;114:209–17.
- [90] Elminshawy NAS, Mohamed AMI, Morad K, Elhenawy Y, Alrobaian AA. Performance of PV panel coupled with geothermal air cooling system subjected to hot climatic. Appl Therm Eng 2019;148:1–9.
- [91] Jakhar S, Soni MS, Gakkhar N. An integrated photovoltaic thermal solar (IPVTS) system with earth water heat exchanger cooling: energy and exergy analysis. Sol Energy 2017;157:81–93.
- [92] Yildiz A, Ozgener O, Ozgener L. Exergetic performance assessment of solar photovoltaic cell (PV) assisted earth to air heat exchanger (EAHE) system for solar greenhouse cooling. Energy Build 2011;43(11):3154–60.
- [93] Zapałowicz Z, Opiela A. Boundary value of the air distribution coefficient that ensures working effectivity of the air-condition system connected with ground heat exchanger and with PV installation. Sustain Cities Soc 2018;42:93–9.
- [94] Bharadwaj SS, Bansal NK. Temperature distribution inside ground for various surface conditions. Build Environ 1981;16(3):183–92.
- [95] Mihalakakou G, Santamouris M, Asimakopoulos D, Argiriou A. On the ground temperature below buildings. Sol Energy 1995;55(5):355–62.
- [96] Florides G, Kalogirou S. Ground heat exchangers-A review of systems, models and applications. Renew Energy 2007;32(15):2461–78.
- [97] Asghar AH, Sharma AK, Singh SP. Effect of different earth surface treatments on sub-soil temperatures. Int J Energy Res 1990;14(9):911–24.
- [98] Popiel CO, Wojtkowiak J, Biernacka B. Measurements of temperature distribution in ground. Exp Therm Fluid Sci 2001;25(5):301–9.
- [99] Sodha MS, Bansal NK, Seth AK. Variance of the ground temperature distribution. Appl Energy 1981;8(4):245–54.

G. Mihalakakou et al.

- [100] Ghosal MK, Tiwari GN, Srivastava NSL, Sodha MS. Thermal modelling and experimental validation of ground temperature distribution in greenhouse. Int J Energy Res 2004;28(1):45–63.
- [101] Ben Jmaa Derbel H, Kanoun O. Investigation of the ground thermal potential in Tunisia focused towards heating and cooling applications. Appl Therm Eng 2010; 30(10):1091–100.
- [102] Mihalakakou G, Santamouris M, Lewis JO, Asimakopoulos DN. On the application of the energy balance equation to predict ground temperature profiles. Sol Energy 1997;60(3–4):181–90.
- [103] Carslaw H, Jaeger J. Conduction of heat in solids. second ed. Oxford: Oxford Science Publishers; 1980.
- [104] Jacovides CP, Mihalakakou G, Santamouris M, Lewis JO. On the ground temperature profile for passive cooling applications in buildings. Sol Energy 1996;57(3):167–75.
- [105] Mihalakakou G. On estimating soil surface temperature profiles. Energy Build 2002;34(3):251–9.
- [106] Popiel CO, Wojtkowiak J. Temperature distributions of ground in the urban region of Poznan City. Exp Therm Fluid Sci 2013;51:135–48.
- [107] Chalhoub M, Bernier M, Coquet Y, Philippe M. A simple heat and moisture transfer model to predict ground temperature for shallow ground heat exchangers. Renew Energy 2017;103:295–307.
- [108] Ahmed SF, Amanullah MTO, Khan MMK, Rasul MG, Hassan NMS. Parametric study on thermal performance of horizontal earth pipe cooling system in summer. Energy Convers Manag 2016;114:324–37.
- [109] Krarti M, Kreider JF. Analytical model for heat transfer in an underground air tunnel. Energy Convers Manag 1996;37(10):1561–74.
- [110] Wu Y, Gan G, Verhoef A, Vidale PL, Gonzalez RG. Experimental measurement and numerical simulation of horizontal-coupled slinky ground source heat exchangers. Appl Therm Eng 2010;30(16):2574–83.
- [111] Spalding D. PHOENICS documentation. Version 2.2. CHAM, London, UK: "; 1995.
 [112] "ANSYS. FLUENT user's guide," 2006 [Online]. Available: http://aerojet.engr.ucd
- avis.edu/fluenthelp/html/ug/node998.htm. [113] Chakraborty K, Mehrotra K, Mohan CK, Ranka S. Forecasting the behavior of
- multivariate time series using neural networks. Neural Network Nov. 1992;5(6): 961–70.
- [114] Rumelhart ADE, Hinton AGE, Williams ARJ. Learning internal representations by error propagation. " in *Parallel distributed processing: explorations in the* microstructure of cognition, vol. 1. foundations; 1986. p. 318–62.
- [115] Trombe A, Pettit M, Bourret B. Air cooling by earth tube heat exchanger: experimental approach. Renew Energy 1991;1(5–6):699–707.
- [116] Ozgener L, Ozgener O. An experimental study of the exergetic performance of an underground air tunnel system for greenhouse cooling. Renew Energy 2010;35 (12):2804–11.
- [117] Vaz J, Sattler MA, Brum RDS, Dos Santos ED, Isoldi LA. An experimental study on the use of Earth-Air Heat Exchangers (EAHE). Energy Build 2014;72:122–31.
- [118] Mongkon S, Thepa S, Namprakai P, Pratinthong N. Cooling performance and condensation evaluation of horizontal earth tube system for the tropical greenhouse. Energy Build 2013;66:104–11.

Renewable and Sustainable Energy Reviews 155 (2022) 111921

- [119] Elminshawy NAS, Siddiqui FR, Farooq QU, Addas MF. Experimental investigation on the performance of earth-air pipe heat exchanger for different soil compaction levels. Appl Therm Eng 2017;124:1319–27.
- [120] Yusof TM, Ibrahim H, Azmi WH, Rejab MRM. Thermal analysis of earth-to-air heat exchanger using laboratory simulator. Appl Therm Eng 2018;134:130–40.
- [121] Wei H, Yang D, Wang J, Du J. Field experiments on the cooling capability of earth-to-air heat exchangers in hot and humid climate. Appl Energy 2020;276: 115493.
- [122] Mihalakakou G, Santamouris M, Asimakopoulos D. On the cooling potential of earth to air heat exchangers. Energy Convers Manag 1994;35(5):395–402.
- [123] Mihalakakou G, Lewis JO, Santamouris M. On the heating potential of buried pipes techniques - application in Ireland. Energy Build 1996;24(1):19–25.
- [124] Mihalakakou G, Lewis JO, Santamouris M. The influence of different ground covers on the heating potential of earth-to-air heat exchangers. Renew Energy 1996;7(1):33–46.
- [125] Santamouris M, Mihalakakou G, Balaras CA, Argiriou A, Asimakopoulos D, Vallindras M. Use of buried pipes for energy conservation in cooling of agricultural greenhouses. Sol Energy 1995;55(2):111–24.
- [126] Santamouris M, Mihalakakou G, Balaras CA, Lewis JO, Vallindras M, Argiriou A. Energy conservation in greenhouses with buried pipes. Energy 1996;21(5): 353–60.
- [127] Abbaspour-Fard MH, Gholami A, Khojastehpour M. Evaluation of an earth-to-air heat exchanger for the north-east of Iran with semi-arid climate. Int J Green Energy 2011;8(4):499–510.
- [128] Ascione F, Bellia L, Minichiello F. Earth-to-air heat exchangers for Italian climates. Renew Energy 2011;36(8):2177–88.
- [129] Soni SK, Pandey M, Bartaria VN. Ground coupled heat exchangers: a review and applications. Renew Sustain Energy Rev 2015;47.
- [130] Chel A, Tiwari GN. Performance evaluation and life cycle cost analysis of earth to air heat exchanger integrated with adobe building for New Delhi composite climate. Energy Build 2009;41(1):56–66.
- [131] Bisoniya TS, Kumar A, Baredar P. Energy metrics of earth-air heat exchanger system for hot and dry climatic conditions of India. Energy Build 2015;86: 214–21.
- [132] Li H, Yu Y, Niu F, Shafik M, Chen B. Performance of a coupled cooling system with earth-to-air heat exchanger and solar chimney. Renew Energy 2014;62:468–77.
- [133] Li H, Ni L, Liu G, Yao Y. Performance evaluation of Earth to Air Heat Exchange (EAHE) used for indoor ventilation during winter in severe cold regions. Appl Therm Eng 2019;160:114111.
- [134] Díaz-Hernández HP, et al. Experimental study of an earth to air heat exchanger (EAHE) for warm humid climatic conditions. Geothermics 2020;84:101741.
- [135] Nemati N, Omidvar A, Rosti B. Performance evaluation of a novel hybrid cooling system combining indirect evaporative cooler and earth-air heat exchanger. Energy 2021;215:119216.
- [136] Ahmadi S, Irandoost Shahrestani M, Sayadian S, Maerefat M, Haghighi Poshtiri A. Performance analysis of an integrated cooling system consisted of earth-to-air heat exchanger (EAHE) and water spray channel. J Therm Anal Calorim 2021;143 (1):473–83.
- [137] Assael MJ, Wakeham WA. In: Letcher TM, editor. "Transport properties and industry," in chemical Thermodynamics and industry. Springer-Verlag; 2004.